MAR. 20. 2007 5:19PM

(3) FISH & RICHARDSON 6175428906

NO. 9382 P. 2

Substitute Form PTO-1595 Attorney Docket No.: 10591-003008

# RECORDATION FORM COVER SHEET PATENTS ONLY

	ot/s) or conv(ies)	
Commissioner for Patents: Please record the attached original docume	2. Name and address of receiving party(ies):	
1. Name of conveying party(les):		
Lernout & Hauspie Speech Products N.V.	Scott L. Baena, Plan Administrator for Post Effective Date L&H	
Flanders Language Valley 50	Bilzin, Sumberg, Baena, Price & Axelrod LLP	
B-8900 leper	200 S. Biscayne Boulevard, Suite 2500	
BELGIUM	Miami, FL 33131	
Additional name(s) attached? 🛛 Yes 🗵 No		
3. Nature of conveyance:		
□ Assignment		
Merger     Security Agreement	1	
Change of Name		
Other Official Committee of Unsecured Creditors of		
Lernout & Hauspie Speech Products N.V.'s Plan of Liquidation for Lernout & Hauspie Speech Products N.V.		
Under Chapter 11 Of The Bankruptcy Code		
Execution Date: March 11, 2003	Additional names/addresses attached?  Yes  No	
4. Application number(6) or patent number(s):		
If this document is being filed with a new application, the execution	n date of the application is:	
A. Patent Application No(s).:	B: Patent No(s).:	
	5,377,303	
09/783,725	0,011,000	
09/852,049		
10/102,047	l attached? □ Yes ⊠ No	
<ol><li>Name/address of party to whom correspondence concerning document should be mailed:</li></ol>	6. Total number of applications/patents involved: 1	
DAVID L. FEIGENBAUM	7. Total fee (37 CFR §3.41): \$160	
Fish & Richardson P.C.		
225 Franklin Street	Authorized to charge Deposit Account.	
Boston, MA 02110	8. Deposit Account No.: 06-1050	
	Please apply any additional charges, or any credits, to our Deposit Account No. 06-1050.	
DO NOT US	E THIS SPACE	
A Statement and Signature: To the best of my knowledge ar	nd belief, the foregoing information is true and correct and	
any attached copy is a true copy of the original docume	ent.	
	) [	
David L. Feigenbaum	- 3/100/7	
Reg. No. 30,378		
Name of Person Signing Signature		
	al number of pages including coversheet, attachments and document: 68	
21588878.doc CERTIFICATE OF TRANSMISSION	BY FACSIMILE	
I keeply artify that this correctondence i	s being transmitted by facsimile to the Patent and Trademark Office on the date	

I hereby certify that this correspondence is being transmitted by facsimile to the Patent and Transmark Ornee on the date indicated below.

March 20, 2007 Date of Transmission

Signature

Typed Name of Person Signing Certificate PATENT REEL: 019047 FRAME: 0157 MAR. 20. 2007 5:19PM (3) FISH & RICHARDSON 6175428906

### IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

Chapter 11

Case No. 00-4398 (JHW)

	х
In re:	;
	;
LERNOUT & HAUSPIE SPEECH	1
PRODUCTS N.V.,	:
,	;
Debtor.	:
	x

OFFICIAL COMMITTEE OF UNSECURED CREDITORS OF LERNOUT & HAUSPIE SPEECH PRODUCTS N.V.'S PLAN OF LIQUIDATION FOR LERNOUT & HAUSPIE SPEECH PRODUCTS N.V. <u>UNDER CHAPTER 11 OF THE BANKRUPTCY CODE</u>

### AKIN GUMP STRAUSS HAUER & FELD LLP

590 Madison Avenue New York, New York 10022-2524 (212) 872-1000

— anđ —

## MONZACK & MONACO, P.A.

1201 Orange Street Wilmington, DE 19801 (302) 656-8162

Attorneys for the Official Committee of Unsecured Creditors of Lemout & Hauspie Speech Products N.V.

Dated: March 11, 2003

# TABLE OF CONTENTS

SECTION H	DEFINITIONS	
1.1	"Administrative Expense Claim"	1
1.2	"Affiliate"	]
1.3	"Allowed"	1
1.4	"Allowed Unsecured Claim Trust Interests"	2
1.5	"Assigned Causes of Action"	2
1.6	"Assumption And Assignment Schedule"	2
1.7	"Available Cash"	.,.Z
1.8	"Avoidance Actions"	3
1.9	"Ballot"	3
1.10	"Bankniptcy Code"	3
1.11	"Bankruptcy Court"	3
1,12	"Bankruptev Rules"	3
1.13	"Belgian Assets"	3
1.14	"Belgian Case"	3
1.15	"Belgian Claims",	3
1.16	"Belgian Priority Claims"	Ξ
1.17	"Belgian Revolving Credit Facility"	4
1,18	"Business Day"	4
J.19	"Cash"	4
1.20	"Cash Investment Yield"	4
1.21	"Causes of Action"	4
1.22	"Chapter 11 Assets"	4
1.23	"Chapter 11 Case"	4
1.24	"Claim"	<b>&gt;</b>
1.25	"Class"	5 F
1.26	"Collateral"	ې ح
1.27	"Committee"	с
1.28	"Common Stock"	Э с
1.29	"Confirmation Date"	
1,30	"Confirmation Hearing"	с
1.31	"Confirmation Order"	с г
1.32	"Curators"	с с
1.33	"Cure"	רייי בייי
1.34	"Debtor"	כ ک
1,35	"Debtor in Possession"	0 2
1.36	"Dictaphone"	0 ک
1.37	"Dictaphone Guaranty"	0 م
1.38`	"Dictaphone Plan"	0 ک
1.39	"Disallowed Claim"	ט, א
1.40	"Disclosure Statement"	۰۰۰۰ ۲
1.41	"Disclosure Statement Approval Date"	u ∠
1.42	"Disclosure Statement Approval Order"	0
1,43	"Disputed Claim"	0

ì

# (3) FISH & RICHARDSON 6175428906

1.44	"Disputed Claims Reserve"	6
1.45	"Distributions"	7
1.46	"Effective Datc"	7
1.47	"Entity"	7
1.48	"Equity Interest"	7
1.49	"Estate"	7
	"Excess Available Cash"	7
1.50	"Final Distribution Date"	
1,51	"Final Order"	
1.52	"Indenture Trustees' Fces"	Ŕ
1.53	"Intercompany Loan Agreement"	
1.54	"Intercompany Loan Agreement Claims"	\$
1,55	"Intercompany Loan Agreement Claims	g
1.56	"L&H Holdings"	Q
1.57	"L&H Holdings Plan"	Q
1.58	"L&H NV"	0
1.59	"L&H NV Parties"	а А
1.60	"Lenders"	ш.У О
1.61	"Liabilities"	я О
1.62	"Lien"	9
1.63	"Litigation Administrative Costs"	9
1.64	"Litigation Proceeds"	
1.65	"Litigation Trust"	У
1.66	"Litigation Trust Agreement"	y
1.67	"Litigation Trust Beneficiary"	
1.68	"Litigation Trustee"	9
1.69	"Litigation Trust Reserve"	9
1,70	"Litigation Trust Beneficial Interest"	9
1.71	"Litigation Monitoring Committee"	9
1.72	"Litigation Monitoring Committee Reserve"	10
1.73	"Litigation Reserve"	.10
1.74	"Maximum Recovery Amount"	10
1.75	"Old Capital Stock"	10
1.76	"Old Convertible Subordinated Notes"	. 10
1.77	"Old Convertible Subordinated Notes Claims"	10
1.78	"Old Convertible Subordinated Notes Indenture"	10
1.79	"Old Convertible Subordinated Notes Indenture Trustee"	.10
1.80	"Old Stock Options"	.10
1.81	"Person"	10
1.82	"Petition Date",	10
1.83	"PIERS Indenture Trustee"	.10
1.84	"PTERS/Old Convertible Subordinated Notes Claims"	.10
1.85	"PIERS Transaction Claims"	.,11
1.86	"Plan"	
1.87	"Plan Administrator"	.11
1.88	"Plan Administration Agreement"	
1.89	"Plan Supplement"	11
a - 14 - 1		



1.90	"Plan Transactions"	11
1.91	"Post Effective Date L&H"	11
1.92	"Priority Non-Tax Claim"	11
1.93	"Priority Tax Claim"	11
1.94	"Professional Fees"	, 11
1.95	"Ratable Proportion"	12
1.96	"Record Date"	12
1.95	"Reinstated" or "Reinstatement"	12
1.98	"Reorganized Dictaphone"	12
1.99	"Schedules"	12
1.100	"Semi-Annual Distribution Date"	12
1.103	"Secured Claim"	,12
1.102	"Securities Aot"	13
1.102	"Securities Class Actions/Suits"	13
1.104	"Securities Law Claim"	13
1.104	"Surplus Proceeds"	13
1.106	"Transfer",	13
1.107	"United States Trustee"	13
1.108	"Unsecured Claim"	13
1.109	"Unsecured Deficiency Claim"	13
1.110	"Voting Deadline"	14
	STRUCTION AND COMPUTATION OF TIME	
SECTION PRIORII	TH PAYMENT OF ADMINISTRATIVE EXPENSE CLAIMS,	14
PRIORIT	III PAYMENT OF ADMINISTRATIVE EXPENSE CLAIMS, TY TAX CLAIMS AND OTHER UNCLASSIFIED CLAIMS	14
SECTION PRIORIT 3.I 3.2	III PAYMENT OF ADMINISTRATIVE EXPENSE CLAIMS, TY TAX CLAIMS AND OTHER UNCLASSIFIED CLAIMS	14
<b>PRIORIT</b> 3.1 3.2	TH PAYMENT OF ADMINISTRATIVE EXPENSE CLAIMS, FY TAX CLAIMS AND OTHER UNCLASSIFIED CLAIMS Administrative Expense Claims Priority Tax Claims	14 15
PRIORIT 3.I 3.2 SECTION IN	TH       PAYMENT       OF       ADMINISTRATIVE       EXPENSE       CLAIMS,         FY TAX CLAIMS AND OTHER UNCLASSIFIED CLAIMS       Administrative Expense Claims       Administrative Expense Claims         Administrative Expense Claims       Priority Tax Claims       Administrative Expense Claims         V CLASSIFICATION OF CLAIMS AND EQUITY INTERESTS       Administrative Expense	14 15 15
PRIORIT 3.1 3.2 SECTION IN 4.1	III PAYMENT OF ADMINISTRATIVE EXPENSE CLAIMS,         IY TAX CLAIMS AND OTHER UNCLASSIFIED CLAIMS         Administrative Expense Claims         Priority Tax Claims         V CLASSIFICATION OF CLAIMS AND EQUITY INTERESTS         Support	14 15 15 15
PRIORIT 3.1 3.2 SECTION IV 4.1 4.1.1	III PAYMENT OF ADMINISTRATIVE EXPENSE CLAIMS, TY TAX CLAIMS AND OTHER UNCLASSIFIED CLAIMS Administrative Expense Claims Priority Tax Claims V CLASSIFICATION OF CLAIMS AND EQUITY INTERESTS Summary	14 15 15 15 
PRIORIT 3.1 3.2 SECTION IN 4.1	III PAYMENT OF ADMINISTRATIVE EXPENSE CLAIMS,         IY TAX CLAIMS AND OTHER UNCLASSIFIED CLAIMS         Administrative Expense Claims         Priority Tax Claims         V CLASSIFICATION OF CLAIMS AND EQUITY INTERESTS         Summary	14 15 15 15 16
PRIORIT 3.1 3.2 SECTION IV 4.1 4.1.1 4.2	<ul> <li>HI PAYMENT OF ADMINISTRATIVE EXPENSE CLAIMS, TY TAX CLAIMS AND OTHER UNCLASSIFIED CLAIMS</li></ul>	14 15 15 16 16
PRIORIT 3.1 3.2 SECTION IV 4.1 4.1.1 4.2 4.2.1	III PAYMENT OF ADMINISTRATIVE EXPENSE CLAIMS, TY TAX CLAIMS AND OTHER UNCLASSIFIED CLAIMS	14 15 15 16 16
PRIORIT 3.1 3.2 SECTION IV 4.1 4.1.1 4.2 4.2.1 4.2.2	<ul> <li>HI PAYMENT OF ADMINISTRATIVE EXPENSE CLAIMS, FY TAX CLAIMS AND OTHER UNCLASSIFIED CLAIMS</li></ul>	14 15 15 16 16 16
PRIORIT 3.1 3.2 SECTION IV 4.1 4.1.1 4.2 4.2.1 4.2.2 4.2.3	<ul> <li>III PAYMENT OF ADMINISTRATIVE EXPENSE CLAIMS,</li> <li>IY TAX CLAIMS AND OTHER UNCLASSIFIED CLAIMS</li></ul>	14 15 15 16 16 16 16
PRIORIT 3.1 3.2 SECTION IN 4.1 4.1.1 4.2 4.2.1 4.2.2 4.2.3 4.2.4	<ul> <li>III PAYMENT OF ADMINISTRATIVE EXPENSE CLAIMS,</li> <li>IY TAX CLAIMS AND OTHER UNCLASSIFIED CLAIMS</li></ul>	14 15 15 16 16 16 16 16
PRIORIT 3.1 3.2 SECTION IV 4.1 4.1.1 4.2 4.2.1 4.2.2 4.2.3 4.2.4 4.2.5	<ul> <li>III PAYMENT OF ADMINISTRATIVE EXPENSE CLAIMS, TY TAX CLAIMS AND OTHER UNCLASSIFIED CLAIMS</li></ul>	14 15 15 16 16 16 16 16 16
PRIORIT 3.1 3.2 SECTION IV 4.1 4.1.1 4.2 4.2.1 4.2.2 4.2.3 4.2.4 4.2.5 4.2.5 4.2.6	<ul> <li>IH PAYMENT OF ADMINISTRATIVE EXPENSE CLAIMS, TY TAX CLAIMS AND OTHER UNCLASSIFIED CLAIMS</li></ul>	14 15 15 16 16 16 16 16 16 16
PRIORIT 3.1 3.2 SECTION IV 4.1 4.1.1 4.2 4.2.1 4.2.2 4.2.3 4.2.4 4.2.5 4.2.5 4.2.6	<ul> <li>III PAYMENT OF ADMINISTRATIVE EXPENSE CLAIMS, TY TAX CLAIMS AND OTHER UNCLASSIFIED CLAIMS</li></ul>	14 15 15 16 16 16 16 16 16 16
PRIORIT 3.1 3.2 SECTION IV 4.1 4.1.1 4.2 4.2.1 4.2.2 4.2.3 4.2.4 4.2.5 4.2.6 4.2.7 SECTION V	III PAYMENT OF ADMINISTRATIVE EXPENSE CLAIMS,         IY TAX CLAIMS AND OTHER UNCLASSIFIED CLAIMS	14 15 15 16 16 16 16 16 16 16 16
PRIORIT 3.1 3.2 SECTION IV 4.1 4.1.1 4.2 4.2.1 4.2.2 4.2.3 4.2.4 4.2.5 4.2.6 4.2.7 SECTION V	III PAYMENT OF ADMINISTRATIVE EXPENSE CLAIMS,         IY TAX CLAIMS AND OTHER UNCLASSIFIED CLAIMS	14 15 15 16 16 16 16 16 16 16 16
PRIORIT 3.1 3.2 SECTION IV 4.1 4.1.1 4.2 4.2.1 4.2.2 4.2.3 4.2.4 4.2.5 4.2.6 4.2.7 SECTION V	IH       PAYMENT OF ADMINISTRATIVE EXPENSE CLAIMS,         IY TAX CLAIMS AND OTHER UNCLASSIFIED CLAIMS	

5.2	Claims Against the Chapter 11 Assets and Equity Interests in L&H NV	.17
5.2.1	Class 1: Priority Non-Tax Claims	,17
5.2.2	Class 2: Secured Claims	.17
5.2.3	Class 3: Unsecured Claims.	.17
5.2.5 5.2.4	Class 4: PIERS/Old Convertible Subordinated Notes Claims	18
•	Class 5: Common Stock	12
5.2.5		40
5.2.6	Class 6: Securities Law Claims.	
5.2.7	Class 7: Other Equity Interests	.10
SECTION V	I IMPAIRED AND UNIMPAIRED CLASSES OF CLAIMS AND	
EQUITY	INTERESTS UNDER THE PLAN; ACCEPTANCE OR REJECTION	
<b>OFTHE</b>	PLAN	19
6.1	Holders of Claims and Equity Interests Entitled to Vote	
6.2	Acceptance by Unimpaired Classes	10
	Elimination of Vacant Classes	10
6,3	Non-consensual Confirmation	
6.4		
6.5	Revocation of the Plan	.17
SECTION VI	II MEANS OF IMPLEMENTATION OF THE PLAN	<b>Z</b> 0
7.1	Transactions on the Effective Date	.20
7.2	Approval of Settlements	.20
7.2.1	Periodic Distributions of Available Cash and Litigation Trust Beneficial	
1.2.1	Interests	20
7.3	Post Effective Date L&H	
7.5 7.4	The Litigation Trust.	
	The Establishment of the Litigation Trust	
7.4.1		
7.4.2	Acquisition of Assigned Causes of Action	10 10
7.4.3	Establishment of Litigation Trust Beneficial Interest Registers.	.20
7.4.4	Litigation Trust Beneficial Interests Granted on Account of Disputed	~~
	Claims	.43
7.4.5	Limitations on Transferability of Litigation Trust Beneficial Interests	.25
7.4.6	Transfer of Causes of Action by Holders	.23
7.4.7	Funding the Litigation Trust	.23
7.4.8	Application of Proceeds and Expenses	.24
7.4.9	Distribution of Litigation Proceeds	.24
7.4.10	Distribution by Litigation Trustee	25
7.4.11	Litigation Reserve	.25
7.4.12	Distributions After Disallowance	,25
	Term	
7.4.14	Powers and Dutics of the Litigation Trustee.	25
7415	Authority to Settle Causes of Action	.26
7,7,15	Compensation of the Litigation Trustee	.27
1.4.10	Indemnification and Exculpation of the Litigation Trustee and Litigation	
1.4.11	The second and provide and of the present transferring the Second	77
	Monitoring Committee	-21 30
7.4.18	Monitoring of Litigation Trust	

•

.

7	.4.19	Discretion Afforded to Litigation Monitoring Committee to Modify Terms of Compensation of Litigation Trustee	29
-		Cancellation of Equity Interests	.29
	.5	Operations of L&H NV Between Confirmation and the Effective Date	29
-		Closing of Chapter 11 Case	29
	.7	Exclusivity Period	29
-	.8	Exclusivity Period	29
-	.9	Committee	
•		Effectuating Documents; Further Transactions	
-	.11	Assumptions of Liabilities	. 30
,	.12	Assumptions of Liabilities	.30
	.13	Preservation of Certain Causes of Action	.30
	,14	Preservation of Centain Causes of Action	30
	.15	Cancellation of Existing Securities	30
7	.16	Appointment of Scott L. Baena to Administer Plan	31
7	.16.1	Distributions Under Plan	21
		Plan Administrator's Obligations Under Plan	21
	7.17	Securities Law Matters	
SECTIC	N VI	II DISTRIBUTIONS UNDER THE PLAN,	. 32
8	1.1	Timing of Distributions,	.32
8	3.2	Delivery of Distributions	
8	E.3	Record Date.	
8	8.3.1	Fruity Interests	
8	3.3.2	Record Date for Holders of Claims	32
8	3.4	Time Bar to Cash Payments by Check.	
8	3.5	Nicested Claims Reserves	
5	3.6	Tay Requirements for Income Generated by Disputed Claims Reserves	دد
Ε	3.7	Untimely Claims	<i>&gt;&gt;</i>
ε	3.8	Estimation of Claims	
8	3.9	Distributions After Effective Date	
8	3.10	Fractional Shares	
8	3,11	Reactional Cents	
8	3.12	Minimum Distribuions	
Ę	3.13	Interest on Claims	34
8	3.14	No Distribution in Excess of Allowed Amount of Claim	
8	8.15	Setoffs	
8	8.16	Payment of Taxes on Distributions Received Pursuant to Plan	دد
٤	B.17	Allocation of Distributions	35
SECTION	ON IX	DISPUTED CLAIMS UNDER THE PLAN	35
	9.1	Objection Deadbare	.,.35
-	9.2	Prosecution of Objections after Effective Date	
	9.3	No Distributions Pending Allowance	, SO
	9.4	Withholding of Allocated Distributions	
		Distribution When a Disputed Claim Becomes an Allowed Claim	36
ţ	9.5	Distribution when a Disputed Claim Becomes an Allowoo Claim	

۷

L ELE: E TRU	EXECUTORY CONTRACTS AND UNEXPIRED LEASES UNDER	) <b>j</b> . n. op
10.1	All Executory Contracts and Unexpired Leases Rejected if Not Listed on	
40.1	Assumption Schedule	
10.2	Assumed Executory Contracts and Unexpired Leases	
10.2	Payments Related to Assumption of Executory Contracts and Unexpired	
(V.)	Leases	
10.4	Bar Date for Rejection Damages	
10.5	Retirec Banefits	
SECTION X	I CONDITIONS PRECEDENT TO THE CONFIRMATION DATI	E
AND THE	E EFFECTIVE DATE	
11.1	Conditions Precedent to the Confirmation of Plan	
11.2	Conditions Precedent to the Effective Date of the Plan	•••
1121	Confirmation Order	
' 11.2.1	Conditions to the Confirmation Date Remain Satisfied.	
11 2 2	Execution of Documents	
11.2	Release of Collateral.	• • •
11.3	Waiver of Conditions Procedent.	
11.2	Effect of Failure or Absence of Waiver of Conditions Precedent to the	
11.0.1	Effective Date of the Plan.	
SECTION X	II EFFECT OF CONFIRMATION	
12.1	Post Effective Date L&H's Authority	
12.2	Vesting and Lights.	
12.3	Injunction	
16.7	Term of Injunctions or Stays	
12.4	Territ of fightledoup of budys and inter-	
12.4	III EXCULPATION	
12.4	III EXCULPATION	
12.4 SECTION XI 13.1	III EXCULPATION	
12.4 SECTION X	III EXCULPATION	
12.4 SECTION XI 13.1 13.2 13.3	III EXCULPATION Avoidance and Recovery Actions Exculpation Release Of Certain Intercompany Claims IV RETENTION OF JURISDICTION	
12.4 SECTION XI 13.1 13.2 13.3	III EXCULPATION Avoidance and Recovery Actions Exculpation Release Of Certain Intercompany Claims IV RETENTION OF JURISDICTION Retention of Jurisdiction	
12.4 SECTION XI 13.1 13.2 13.3 SECTION XI	Avoidance and Recovery Actions Exculpation Release Of Certain Intercompany Claims	 
12.4 SECTION XI 13.1 13.2 13.3 SECTION XI 14.1 14.2	III EXCULPATION         Avoidance and Recovery Actions         Exculpation         Release Of Certain Intercompany Claims         IV RETENTION OF JURISDICTION         Retention of Jurisdiction         Modification of Plan.         V MISCELLANEOUS PROVISIONS	
12.4 SECTION XI 13.1 13.2 13.3 SECTION XI 14.1 14.2	III EXCULPATION         Avoidance and Recovery Actions         Exculpation         Release Of Certain Intercompany Claims         IV RETENTION OF JURISDICTION         Rotention of Jurisdiction         Modification of Plan         V MISCELLANEOUS PROVISIONS         Payment of Statutory Fees	
12.4 SECTION XI 13.1 13.2 13.3 SECTION XI 14.1 14.2 SECTION X	III EXCULPATION         Avoidance and Recovery Actions         Exculpation         Release Of Certain Intercompany Claims         IV RETENTION OF JURISDICTION         Rotention of Jurisdiction         Modification of Plan         V MISCELLANEOUS PROVISIONS         Payment of Statutory Fees         Role of Indenture Trustees; Certain Fees And Expenses of Indenture	
12.4 SECTION XI 13.1 13.2 13.3 SECTION XI 14.1 14.2 SECTION X 15.1	III EXCULPATION         Avoidance and Recovery Actions         Exculpation         Release Of Certain Intercompany Claims         IV RETENTION OF JURISDICTION         Retention of Jurisdiction         Modification of Plan.         V MISCELLANEOUS PROVISIONS         Payment of Statutory Fees         Role of Indenture Trustees; Certain Fees And Expenses of Indenture         Trustees	
12.4 SECTION XI 13.1 13.2 13.3 SECTION XI 14.1 14.2 SECTION X 15.1	III EXCULPATION         Avoidance and Recovery Actions         Exculpation         Release Of Certain Intercompany Claims         IV RETENTION OF JURISDICTION         Rotention of Jurisdiction         Modification of Plan         V MISCELLANEOUS PROVISIONS         Payment of Statutory Fees         Role of Indenture Trustees; Certain Fees And Expenses of Indenture         Trustees         Governing Law	
12.4 SECTION XI 13.1 13.2 13.3 SECTION XI 14.1 14.2 SECTION X 15.1 15.2	III EXCULPATION         Avoidance and Recovery Actions         Exculpation         Release Of Certain Intercompany Claims         IV RETENTION OF JURISDICTION         Rotention of Jurisdiction         Modification of Plan         V MISCELLANEOUS PROVISIONS         Payment of Statutory Fees         Role of Indenture Trustees; Certain Fees And Expenses of Indenture         Trustees         Governing Law         Notices	
12.4 SECTION XI 13.1 13.2 13.3 SECTION X 14.1 14.2 SECTION X 15.1 15.2 15.3	III EXCULPATION         Avoidance and Recovery Actions         Exculpation         Release Of Certain Intercompany Claims         IV RETENTION OF JURISDICTION         Retention of Jurisdiction         Modification of Plan.         V MISCELLANEOUS PROVISIONS         Payment of Statutory Fees         Role of Indenture Trustees; Certain Fees And Expenses of Indenture         Trustees	

.

,

15.7	Plan Controls; Conflicts
	Reservation of Rights
	Tax Reporting and Compliance
	Binding Effect



# THE OFFICIAL COMMITTEE OF UNSECURED CREDITORS OF LERNOUT & HAUSPIE SPEECH PRODUCTS N.V.'S PLAN OF LIQUIDATION FOR LERNOUT & HAUSPIE SPEECH PRODUCTS N.V. <u>UNDER CHAPTER 11 OF THE BANKRUPTCY CODE</u>

The Official Committee of Unsecured Creditors of Lernout & Hauspie Speech Products N.V. proposes the following Plan of Liquidation pursuant to the provisions of Chapter 11 of the Bankrupicy Code:

### SECTION I

### DEFINITIONS

The following terms, when used in the Plan or any subsequent amendments or modifications thereof, shall have the meanings defined below:

1.1 "Administrative Expense Claim" means any right to payment constituting a cost or expense of administration of the Chapter 11 Case Allowed under sections 503(b) and 507(a)(1) of the Bankruptcy Code, including, without limitation, (a) any actual and necessary costs and expenses of preserving the Chapter 11 Assets, (b) any actual and necessary costs and expenses of operating the businesses related to the Chapter 11 Assets, (c) any indebtedness or obligations incurred or assumed by the Debtor in the ordinary course of business in connection with the Chapter 11 Assets, (d) claims for reclamation of the Chapter 11 Assets Allowed in accordance with section 546(c)(2) of the Bankruptcy Code pursuant to a Final Order, (e) any Professional Fees, whether fixed before or after the Effective Date, (f) any fees or charges assessed against and payable by the Estate of the Debtor under section 1930, Chapter 123, title 28, United States Code, including post-Confirmation Date and post-Effective Date fees and charges, and (g) the Indenture Trustees' Fees. No Belgian Priority Claim shall constitute an Administrative Expense Claim.

1.2 "<u>Affiliate</u>" shall have the meaning ascribed to such term in section 101(2) of the Bankruptcy Code.

1.3 "<u>Allowed</u>" means, with reference to any Claim (including any Administrative Expense Claim or Belgian Claim) in the Chapter 11 Case, (a) any Claim against or Equity Interest in the Debtor, proof of which was filed within the applicable period of limitation fixed by the Bankruptcy Court in accordance with Rule 3003(c)(3) of the Bankruptcy Rules (i) as to which no objection to the allowance thereof, or action to equitably subordinate or otherwise limit recovery with respect thereto, has been interposed within the applicable period of limitation fixed by the Plan, the Bankruptcy Code, the Bankruptcy Rules or a Final Order, (ii) as to which no action has been commenced to avoid such Claim or Equity Interest within the applicable period of limitation fixed by the Plan, or (iii) as to which an objection has been interposed, to the extent such Claim or Equity Interest has been allowed (whether in whole or in part) by a Final Order, (b) if no proof of Claim was so filed, any Claim against the Debtor which has been listed by the Debtor on its Schedules, as liquidated in amount and not disputed or contingent (or as to which the applicable proof of Claim has been withdrawn or Disallowed), (c)

if no proof of Equity Interest was so filed, any Equity Interest listed in the consolidated stockholders list maintained by L&H NV as of the Record Date, (d) any Claim arising from the recovery of property under sections 550 or 553 of the Bankruptoy Code and allowed in accordance with section 502(h) of the Bankruptoy Code, (e) any Claim allowed under or pursuant to the terms of the Plan, (f) with respect to an Old Convertible Subordinated Notes Claim, or any portion thereof held by any Person or Entity listed on the Old Convertible Subordinated Notes Indenture Trustee's note register in accordance with acctions 2.5(a) and 5.2 of the Old Convertible Subordinated Notes Indenture as of the Record Date, and (g) any Claim to the extent that it has been allowed by a Final Order. Unless otherwise specified herein or by order of the Bankruptcy Court, "Allowed Claims" shall not, for any purpose under the Plan, include interest, penaltics or late charges on such Claims from and after the Petition Date. In addition, "Allowed Claim" shall not include (i) any Claim subject to disallowance in accordance with section 502(d) of the Bankruptcy Code, (ii) any Claim allowed solely for the purpose of voting to accept or reject the Plan pursuant to an order of the Bankruptcy Court or (iii) any Belgian Priority Claim.

1.4 "<u>Allowed Unsecuted Claim Trust Interests</u>" shall have the meaning ascribed to such term in Section 5.2.3(a) of the Plan.

1.5 "Assigned Causes of Action" means collectively, any and all Causes of Action of L&H NV, including, without limitation, any Causes of Action assigned to the Debtor pursuant to the L&H Holdings Plan and pursuant to the Stipulation And Order Between Lernout & Hauspie Speech Products N.V. And L&H Holdings USA, Inc. And Baker Parties Compromising And Settling Claims, dated January 31, 2002, but with respect to all of the foregoing, excluding (i) Causes of Action released in accordance with Section 13.2 of the Plan; (ii) any and all Causes of Action settled or resolved in the Plan or pursuant to a Final Order of the Bankruptey Court; and (iii) any Causes of Action asserted against any third party who has asserted a Claim against the Debtor unless and until (a) the Plan Administrator has asserted any and all of the Debtor's counterclaims against such third party or rights to setoff or recoupment against such third party, including the Debtor's rights under section 502(d) of the Bankruptey Code, and (b) such counterclaims, rights to setoff or recoupment, or rights under section 502(d) of the Bankruptey Code have been determined by a Final Order or otherwise.

1.6 "<u>Assumption And Assignment Schedulc</u>" means the schedule of executory contracts and unexpired leases designated in the Plan Supplement, as may be the case, assumption or assumption and assignment as of the Effective Date of the Plan, pursuant to sections 365 and 1123(b)(2) of the Bankruptcy Code and Sections 10.1 and 10.2 of the Plan.

1.7 "<u>Available Cash</u>" means at any time, the Cash held by Post Effective Date L&H on account of the Chapter 11 Assets (including the net proceeds from any sales of Chapter 11 Assets and any Distributions received, either directly or derivatively, from L&H Holdings pursuant to the L&H Holdings Plan or from Reorganized Dictaphone pursuant to the Dictaphone Plan), along with the non-cash proceeds of any Chapter 11 Asset Transfers; <u>provided</u>, <u>however</u>, that such non-cash proceeds shall include the common stock of Reorganized Dictaphone distributed to L&H NV pursuant to the Dictaphone Plan, *minus* (i) the amount of Cash necessary to satisfy or reserve for all Allowed Class 2 Secured Claims, Administrative Expense Claims,



Priority Tax Claims, Class 1 Priority Non-Tax Claims, and Cash held in any Disputed Claims Reserve, (ii) the amount of Cash determined from time to time by the Plan Administrator to be necessary to fund adequately the administration of the Plan and Post Effective Date L&H on and after the Effective Date, and (iii) the Litigation Trust Reserve and the Litigation Monitoring Committee Reserve.

1.8 "<u>Avoidance Actions</u>" means any and all avoidance or recovery actions under sections 502(d), 542, 544, 545, 547, 548, 549, 550, 551 and 553 of the Bankruptcy Code.

1.9 "<u>Ballot</u>" means the form or forms distributed to each holder of an impaired Claim on which form or forms such holder of a Claim may, among other things, vote to accept or reject the Plan.

1.10 "<u>Bankruptcy Code</u>" means title 11 of the United States Code, as amended from time to time, as applicable to the Chapter 11 Case, as in effect on the Confirmation Date.

1.11 "<u>Bankruptcy Court</u>" means the United States Bankruptcy Court for the District of Delaware having jurisdiction over the Chapter 11 Case and, to the extent of any reference under section 157 of title 28 of the United States Code, the unit of such District Court under section 151 of the United States Code.

1.12 "<u>Bankruptcy Rules</u>" means the Federal Rules of Bankruptcy Procedure as promulgated by the United States Supreme Court under section 2075, title 28, United States Code, and the Local Rules of the Bankruptcy Court, as amended from time to time and applicable to the Chapter 11 Case.

1.13 "<u>Belgian Assets</u>" means the assets attributable to the Debtor's business operations in Belgium as determined by the allocation described in more detail in Section IV.M of the Disclosure Statement plus an additional \$5,091,386 in Cash to be transferred to the Curators for distribution to Belgian Priority Claims.

1.14 "Belgian Case" means the Debtor's bankruptcy proceeding currently pending in Belgium.

1.15 "<u>Belgian Claims</u>" means general, unsecured claims for which proofs of claim (or their functional equivalent under Belgian law) were validly filed in the Belgian Case, which claims will be transferred to the Chapter 11 Case, deemed timely filed therein, and subject to administration and allowance solely and exclusively in the Chapter 11 Case. The priority and validity of each Belgian Claim will be determined solely and exclusively in the Chapter 11 Case in accordance with the Bankruptcy Code and Bankruptcy Rules.

1.16 "<u>Belgian Priority Claims</u>" means claims entitled to priority in the Belgian Case including, but not limited to, (i) the claims which are the subject of the Emergency Motion Of Lemout & Hauspic Speech Products N.V., Pursuant To 11 U.S.C. §§ 105(a), 363(b)(1), And 503(b), For Entry Of Order Authorizing Payment Of Certain Administrative Expense Claims Relating To Post-Petition (A) Statutorily Mandated Employee Payments; (B) Belgian Social Security Taxes; (C) Belgian Employee Withholding Taxes; And (D) Belgian Real Estate Taxes

filed by L&H NV on March 31, 2003, (ii) any claims of a Belgian governmental unit, Rijksdienst voor Sociale Zekerheid, Ministerie Vlaamse Gemeenschap, or Ministerie Van Financien, (iii) any claims for taxes imposed under Belgian law, including, but not limited to, social security tax, real estate tax, income tax, wage tax, employee withholding tax, mandatory severance payments, and (iv) any professional fee claims relating to the Belgian Case, including, but not limited to, fees and expenses of the Curators and the commissioners. The Belgian Priority Claims will not participate or receive any consideration under the Plan. The Belgian Priority Claims will be administered solely and exclusively in the Belgian Case.

1.17 "<u>Belgian Revolving Credit Facility</u>" means that certain Revolving Credit Facility Agreement, dated as of May 2, 2000, by and among the Debtor, as the borrower, and the Lenders, as the lenders.

1.18 "<u>Business Day</u>" means any day other than (i) a Saturday, (ii) a Sunday, (iii) any other day on which banking institutions in New York, New York are required or authorized to close by law or executive order, and (iv) the Friday after Thanksgiving.

1.19 "<u>Cash</u>" means legal tender of the United States of America. When applicable, and except as provided in Section 7.4.8 below, the conversion rate among the United States Dollar (\$) and the Euro ( $\varepsilon$ ) shall be the spot conversion rate expressed in Dollars per Euro, as published at 10 a.m. Eastern Standard Time by the Federal Reserve Bank of New York on the Petition Date.

1.20 "<u>Cash Investment Yield</u>" means the net yield earned by Post Effective Date L&H from the investment of Cash beld pending distribution in accordance with the provisions of the Plan.

1.21 "<u>Causes of Action</u>" means any and all actions, causes of action, liabilities, controversies, promises, agreements, obligations, rights, suita, debts, sums of money, damages, judgments, claims and demands whatsoever, whether known or unknown, reduced to judgment or not reduced to judgment, liquidated or unliquidated, contingent or non-contingent, matured or unmatured, disputed or undisputed, secured or unsecured, assertable directly or derivatively, existing or hereafter arising, in law, equity or otherwise, based in whole or in part upon any act or omission or other event occurring prior to the Petition Date or during the course of the Chapter 11 Case (including through the Effective Date), including, without limitation, the Avoidance Actions and any Causes of Action assigned to the Debtor pursuant to the L&H Holdings Plan.

1.22 "<u>Chapter 11 Assets</u>" means all of the Debtor's assets other than Belgian Assets,

1.23 "<u>Chapter 11 Case</u>" means the case commenced under Chapter 11 of the Bankruptcy Code by the Debtor pending in the Bankruptcy Court, as referenced by Case No. 00-4398 (IHW), which is consolidated administratively under the same Case Number with the chapter 11 cases of Dictaphone (No. 00-4397 (JHW)) and L&H Holdings (No. 00-4399 (JHW)).



1.24 "<u>Claim</u>" means a claim against the Chapter 11 Assets, whether or not asserted, as defined in section 101(5) of the Bankruptcy Code.

1.25 "<u>Class</u>" means a category of holders of Claims or Equity Interests described in Section IV hereof.

1.26 "<u>Collatoral</u>" means any Chapter 11 Asset subject to a Lien to secure the payment or performance of a Claim, which Lien is not subject to avoidance or otherwise invalid under the Bankruptcy Code or other applicable law.

1.27 "<u>Committee</u>" means the official committee of unsecured creditors appointed by the United States Trustee in the chapter 11 cases of L&H NV, Dictaphone, and L&H Holdings on December 13, 2000 and as reconstituted on February 28, 2001 to represent unsecured creditors of L&H NV and L&H Holdings (and, prior to February 28, 2001, also to represent the unsecured creditors of Dictaphone), as such committee may be constituted from time to time. Subsequent to the L&H Holdings Plan becoming effective on September 23, 2002, the Committee only represents the unsecured creditors of L&H NV.

1.28 "<u>Common Stock</u>" means the common stock of L&H NV, with no par value, authorized and outstanding on the Petition Date, including all rights, claims and interests attendant thereto.

1.29 "<u>Confirmation Date</u>" means the date on which the Clerk of the Bankruptcy Court enters the Confirmation Order on the docket of the Bankruptcy Court with respect to the Chapter 11 Case.

1.30 "<u>Confirmation Hearing</u>" means the hearing held by the Bankruptcy Court pursuant to section 1128(a) of the Bankruptcy Code on confirmation of the Plan pursuant to section 1129 of the Bankruptcy Code, as it may be adjourned or continued from time to time.

1.31 "<u>Confirmation Order</u>" means the order of the Bankruptcy Court confirming the Plan under section 1129 of the Bankruptcy Code.

1.32 "<u>Curators</u>" means the Official Receivers, and any receivers under the supervision of the Official Receivers, appointed by the Belgian Court pursuant to the Judgment On Petition of the Belgian Court dated October 24, 2001 and October 25, 2001.

1.33 "Cure" means the Distribution of Cash, or such other property as may be agreed upon by the parties and/or ordered by the Bankruptcy Court, with respect to the assumption of an executory contract or unexpired lease, pursuant to section 365(b) of the Bankruptcy Code, in an amount equal to all accrued, due and unpaid monetary obligations, without interest, or such other amount as may be agreed upon by the parties or ordered by the Bankruptcy Court, under such executory contract or unexpired lease, to the extent such obligations are enforceable under the Bankruptcy Code and applicable non-bankruptcy law.

1.34 "<u>Debtor</u>" means L&H NV.



1.35 "<u>Debtor in Possession</u>" means the Debtor in its capacity as debtor in possession in the Chapter 11 Case under sections 1107(a) and 1108 of the Bankruptcy Code.

1.36 "<u>Diciaphone</u>" means Dictaphone Corporation, a Delaware corporation, formerly a wholly-owned subsidiary of L&H NV and debtor in possession, which emerged from chapter 11 case No. 00-4397 (JHW) on March 28, 2002 to become Reorganized Dictaphone.

1.37 "<u>Dictaphone Guaranty</u>" means that certain Limited Guaranty, dated as of May 30, 2000 and amended and restated on November 8, 2000, of Dictaphone for the benefit of the Lenders.

1.38 "<u>Dictaphone Plan</u>" means the Third Amended Plan Of Reorganization Of Dictaphone Corporation Under Chapter 11 Of Bankruptcy Code filed on January 31, 2002 in the Bankruptcy Court in Dictaphone's chapter 11 case (No. 00-4397 (JHW)) confirmed on March 13, 2002 and effective on March 28, 2002, including the plan supplement, the schedules and the exhibits thereto, either in its present form or as the same may be amended, modified or supplemented from time to time in accordance with the terms thereof.

1.39 "Disallowed Claim" means (a) a Claim, or any portion thereof, that has been disallowed by a Final Order or (b) unless scheduled by the Debtor as a fixed, liquidated, non-contingent and undisputed Claim, a Claim as to which a proof of Claim filing deadline has been established by the Bankruptey Code, Bankruptey Rules or Final Order, but for which no proof of Claim has been timely filed or deemed timely filed with the Bankruptey Court pursuant to either the Bankruptey Code or any Final Order.

1.40 "<u>Disclosure Statement</u>" means the disclosure statement relating to the Plan, including the exhibits and schedules thereto, as the same may be amended, modified or supplemented from time to time, as approved by the Bankruptcy Court pursuant to the Disclosure Statement Approval Order.

1.41 "<u>Disclosure Statement Approval Date</u>" means the date on which the Clerk of the Bankruptcy Court enters the Disclosure Statement Approval Order on the docket of the Bankruptcy Court with respect to the Chapter 11 Case approving the adequacy of the Disclosure Statement under section 1125 of the Bankruptcy Code.

1.42 "<u>Disclosure Statement Approval Order</u>" means the Final Order approving, among other things, the adequacy of the Disclosure Statement pursuant to section 1125 of the Bankruptcy Code, entered by the Bankruptcy Court on April 10, 2003.

1.43 "<u>Disputed Claim</u>" means a Claim, or any portion thereof, that is neither an Allowed Claim nor a Disallowed Claim.

1.44 "<u>Disputed Claims Reserve</u>" means a reserve of Cash, Available Cash, Litigation Trust Beneficial Interests, and/or other Distributions under the Plan, established herein for, among other things, the payment or other satisfaction of Disputed Claims that become Allowed Claims after the Effective Date, which reserve shall be held for the benefit of the



holders of Disputed Claims and, except as provided in Section 8.5 of the Plan, upon closure of a Disputed Claims Reserve, shall not constitute property of the Estate or Post Effective Date L&H.

1.45 "<u>Distributions</u>" means the distribution to be made in accordance with the Plan of, as the case may be: (a) Cash, (b) Litigation Trust Beneficial Interests, (c) Available Cash, and (d) any other consideration distributed to holders of Allowed Claims under the terms and provisions of the Plan.

1.46 "<u>Effective Date</u>" means the first Business Day on which (a) all conditions precedent set forth in Section 11.2 of the Plan have been satisfied or waived as provided in Section 11.3 of the Plan and (b) no stay of the Confirmation Order is in effect.

1.47 "<u>Entity</u>" shall have the meaning assigned to such term in section 101(15) of the Bankruptcy Code.

1.48 "<u>Equity Interest</u>" means, as of the Petition Date, any capital stock or other ownership interest in the Debtor, whether or not transferable, and any option, call, warrant or right to purchase, sell or subscribe for an ownership interest or other equity security in the Debtor, including, but not limited to, (i) the Old Capital Stock of the Debtor and (ii) redemption, conversion, exchange, voting, participation, dividend rights and liquidation, preferences relating to such Old Capital Stock.

1.49 "Estate" means, as to the Debtor and for the purposes of the Plan only, the Chapter 11 Assets.

1.50 "Excess Available Cash" shall have the meaning ascribed to such term in Section 5.2.3 of the Plan.

1.51 "<u>Final Distribution Date</u>" means the final Distribution of Available Cash to holders of Allowed Claims after resolution, abandonment or other disposition of all Disputed Claims, the liquidation into Cash of all of the Chapter 11 Assets, and the collection of other sums due or otherwise remitted or returned to Post Effective Date L&H.

1.52 "Final Order" means an order or judgment of the Banknuptcy Couit as to which the time to appeal, petition for certiorari, or move for reargument or rehearing has expired and as to which no appeal, petition for certiorari or other proceedings for reargument or rehearing shall then be pending, provided, however, if an appeal, petition for certiorari, reargument or rehearing thereof has been filed or sought, such order of the Banknuptcy Court shall have been affirmed by the highest court to which such order was appealed, or certiorari shall have been denied or reargument or rehearing shall have been denied or resulted in no modification of such order, and the time to take any further appeal, petition for certiorari or move for reargument or rehearing shall have expired; provided, further, that the possibility that a motion under Rule 59 or Rule 60 of the Federal Rules of Civil Procedure, or any analogous rule under the Bankruptcy Rules, may be filed with respect to such order shall not cause such order not to be a Final Order.

7





1,53 "Indenture Trustees' Fees" moans the reasonable propertition and postpetition fees and expenses of the PIERS Indenture Trustee and the Old Convertible Subordinated Notes Indenture Trustee and their respective counsel as agreed to by the Plan Administrator or as allowed by the Bankruptcy Court.

1.54 "<u>Intercompany Loan Agreement</u>" means that certain Loan Agreement, dated May 5, 2000, in the amount of \$173 million between Dictaphone, as borrower, and L&H Coordination Centre C.V.B.A., as lender, as amended pursuant to that certain Amended and Restated Intercompany Loan Agreement dated November 20, 2000, and as may be in effect from time to time.

1.55 "Intercompany Loan Agreement Claims" means all Claims against Dictaphone asserted in its chapter 11 case (No. 00-4397 (JHW)) pending in the Bankruptcy Court arising from or under, or relating in any way to, the Intercompany Loan Agreement, including approximately (a) \$173 million in principal amount under such Intercompany Loan Agreement, (b) \$7.9 million in interest under such Intercompany Loan Agreement, and (c) other claims that were treated by Dictaphone in the Dictaphone Plan as being included in such Intercompany Loan Agreement, Claim (although not formally documented under the Intercompany Loan Agreement), including (i) a claim in the amount of \$9.3 million for expenses of L&H NV incurred as a result of the acquisition of Dictaphone, (ii) a claim in the amount of \$3 million for cash advanced to Dictaphone by L&H NV prior to the commencement of the Chapter 11 Case, and (iii) a claim in the amount of \$2.6 million for allocation by L&H NV to Dictaphone of overhead charges.

1.56 "<u>L&H Holdings</u>" means L&H Holdings USA, inc., a Delaware corporation and formerly a debtor and debtor in possession in chapter 11 case No. 00-4399 (JHW) currently pending before the Bankruptcy Court.

1.57 "<u>L&H Holdings Plan</u>" means the First Amended Plan Of Liquidation Of L&H Holdings USA, Inc. Under Chapter 11 Of The Bankruptcy Code, filed in L&H Holdings' chapter 11 case (No. 00-4399 (JHW)) on April 29, 2002, confirmed by the Bankruptcy Court on August 13, 2002, and effective on September 23, 2002.

1.58 "<u>L&H NV</u>" means Lernout & Hauspie Speech Products N.V., a corporation incorporated under Belgian laws as a "naamlooze vennootschap" with registered scat at Flanders Language Valley, 899 Icper, Belgium, registered with the Commercial Registers of Ieper under No. 31.360, Brussels under No. 610.455 and Antwerpen under No. 329.244, that is the Debtor and Debtor-in-Possession in the Chapter 11 Case.

1.59 "<u>L&H NV Parties</u>" means, collectively, (a) the Curators, (b) L&H NV and Fost Effective Date L&H, (c) the Committee and the present and former members thereof, and (d) solely in respect of their specific capacities, such directors, officers, agents, attorneys, affiliates, employees, accountants, advisors and financial advisors of any of the foregoing who, with respect to each of the parties identified in subsections (b) and (c), has served in such capacities on or after November 1, 2001.





1,60 <u>"Londers"</u> means those certain financial institutions acting as the lenders under the Belgian Revolving Credit Facility, and their respective successors or assigns with respect thereto, including Fortis Bank N.V., KBC Bank N.V., Artesia Banking Corporation N.V., Deutsche Bank N.V., and Dresdner Bank Luxembourg S.A.

1.61 "<u>Liabilities</u>" means any and all costs, expenses, actions, Causes of Action, suits, controversies, damages, claims, liabilities or demands of any nature, whether known or unknown, foreseen or unforeseen, existing or hereinafter arising, liquidated or unliquidated, matured or not matured, contingent or direct, whether arising at common law, in equity, or under any statute, based in whole or in part on any act or omission or other occurrence arising or taking place on or prior to the Effective Date relating to the Chapter 11 Assets.

1.62 "<u>Lien</u>" shall have the meaning ascribed to such term in section 101(37) of the Bankruptcy Code (but a lien that has or may be avoided pursuant to an Avoidance Action shall not constitute a Lien).

1.63 "<u>Litigation Administrative Costs</u>" shall have the meaning ascribed to such term in Section 7.4.8 of the Pian.

1.64 "<u>Litigation Proceeds</u>" shall have the meaning ascribed to such term in Section 7.4.9 of the Plan

1.65 <u>"Litigation Trust"</u> means a litigation trust established pursuant to (a) Regulation 301.7701-4(d) of the Regulations of the United States Department of the Treasury and (b) Internal Revenue Procedure 94-45, and as a grantor trust, subject to the provisions of Subchapter J and Subpart E of the Internal Revenue Code of 1986 (as amended), owned by the Trust Beneficiaries as grantors, and to be established in accordance with Section 7.4 of the Plan.

1.66 "<u>Litigation Trust Agreement</u>" means the trust agreement pursuant to which the Litigation Trust shall be established, substantially in the form contained in the Plan Supplement.

1.67 "Litigation Trust Beneficiary" means a beneficiary of the Litigation Trust.

1.68 "<u>Litigation Trustee</u>" means Scott L. Baena, or any successor appointed by the Litigation Monitoring Committee.

1.69 "<u>Litigation Trust Reserve</u>" shall have the meaning ascribed to such term in Section 7.4.7 of the Plan.

1.70 "<u>Litigation Trust Beneficial Interest</u>" means the beneficial interest in the Litigation Trust being distributed to holders of certain Classes of Claims, as described in Section 7.4 of the Plan.

1.71 "<u>Litigation Monitoring Committee</u>" shall have the meaning ascribed to such term in Section 7.4.18 of the Plan.





1.72 "<u>Litigation Monitoring Committee Reserve</u>" shall have the meaning ascribed to such term in Section 7.4.7 of the Plan.

1.73 "<u>Litigation Reserve</u>" shall have the meaning ascribed to such term in Section 7.4.11 of the Plan.

1.74 "<u>Maximum Recovery Amount</u>" shall have the meaning ascribed to such term in Section 7.4.10(b) of the Plan.

1.75 "<u>Old Capital Stock</u>" means, with respect to the Debtor, collectively: (a) the Common Stock of the Debtor and (b) the Old Stock Options of the Debtor.

1.76 "<u>Old Convertible Subordinated Notes</u>" means the 8% Convertible Subordinated Notes Due 2001, issued under that certain Indenture, dated as of November 20, 1996, between L&H NV and the Old Convertible Subordinated Notes Indenture Trustee.

1.77 "<u>Old Convertible Subordinated Notes Claims</u>" means all claims (including, but not limited to, all Claims), directly or indirectly, against the Debtor arising from or under, or relating in any way to, the Old Convertible Subordinated Notes or the Old Convertible Subordinated Notes Indenture.

1.78 "Old Convertible Subordinated Notes Indenture" means that certain Indenture, dated as of November 20, 1996, between L&H NV and the Old Convertible Subordinated Notes Indenture Trustee, as trustee.

1.79 "<u>Old Convertible Subordinated Notes Indenture Trustee</u>" means The Bank of New York in its capacity as successor indenture trustee under the Old Convertible Subordinated Notes Indenture.

1.80 "<u>Old Stock Options</u>" means any options, warrants or other rights to purchase Common Stock of the Debtor, whenever granted.

1.81 "<u>Person</u>" shall have the meaning ascribed to such term in section 101(41) of the Bankniptey Code.

1.82 "<u>Petition Date</u>" means November 29, 2000, the date on which L&H NV filed its chapter 11 petition and commenced the Chapter 11 Case.

1.83 "<u>PIERS Indenture Trustee</u>" means Wilmington Trust Company in its capacity as trustee under (a) that certain Multiple Series Indenture, dated as of May 27, 1998, between L&H NV and Wilmington Trust Company, as trustee, and (b) that certain First Supplemental Indenture, dated 2s of May 27, 1998, between L&H NV and Wilmington Trust Company, as trustee.

1.84 "<u>PIERS/Old Convertible Subordinated Notes Claims</u>" means, collectively, all Old Convertible Subordinated Notes Claims and PIERS Transaction Claims.



10

1.85 "<u>PIERS Transaction Claims</u>" means claims, including but not limited to Claims, against L&H NV based upon, arising from or under, or related to any of the following: (a) that certain Multiple Series Indeature, dated as of May 27, 1998, between L&H NV and Wilmington Trust Company, as trustee, (b) that certain First Supplemental Indeature, dated as of May 27, 1998, between L&H NV and Wilmington Trust Company, as trustee, (c) those certain 4.75% Convertible Junior Subordinated Debentures due 2008, issued by L&H NV, and (d) any guarantee by L&H NV of the 4.75% Preferred Income Equity Securities and 4.75% Common Securities issued under the foregoing agreements, including under that certain Guarantee Agreement, dated as of May 27, 1998, among L&H NV and Wilmington Trust Company.

1.86 "Plan" means the Official Committee Of Unsecured Creditors' Plan Of Liquidation For Lemout & Hauspie Speech Products N.V. Under Chapter 11 Of The Bankruptcy Code, including the Plan Supplement, the schedules (including the Assumption And Assignment Schedule) and exhibits hereto, either in its present form or as the same may be amended, modified or supplemented from time to time in accordance with the terms hereof.

1.87 "<u>Plan Administrator</u>" means Scott L. Baena. All references to Post Effective Date L&H shall include the Plan Administrator.

1.88 "<u>Plan Administration Agreement</u>" means the agreement by which the Plan Administrator shall be appointed, substantially in the form contained in the Plan Supplement.

1.89 "<u>Plan Supplement</u>" means the form of documents specified in Section 15.6 of the Plan, which are incorporated herein by reference.

1.90 "<u>Plan\_Transactions</u>" means the transactions effectuated or to be effectuated on or before the Effective Date pursuant to the terms of the Plan, including, but not limited to, the transactions described in Section VII of the Plan.

1.91 "Post E ffective Date L&H" means, on and after the Effective Date, the successor to L&H NV and the Estate with respect to the Chapter 11 Aasets (and any successor thereto by merger, consolidation or otherwise).

1.92 "<u>Priority Non-Tax Claim</u>" means any Claim of a kind specified in sections 507(a)(3), (4), (5), (6), (7) or (9) of the Bankruptcy Code. No Belgian Priority Claim shall constitute a Priority Non-Tax Claim.

1.93 "<u>Priority Tax Claim</u>" means any Claim of a governmental unit of the kind specified in sections 502(i) and 507(a)(8) of the Bankruptey Code. No Belgian Priority Claim shall constitute a Priority Tax Claim.

1.94 "Professional Fees" means any Claim of a professional, retained in the Chapter 11 Case, pursuant to sections 327, 328 and 1103 of the Bankruptcy Code or otherwise, for compensation or reimbursement of costs and expenses relating to services incurred prior to and including the Effective Date relating to the Chapter 11 Assets, when and to the extent any such Claim is Allowed by the Bankruptcy Court pursuant to sections 328, 330, 331, 503(b), or 1103 of the Bankruptcy Code. 1.95 "<u>Ratable Proportion</u>" means, with reference to any Distribution on account of any Allowed Claim in any Class, the ratio (expressed as a percentage) that the amount of such Allowed C laim b eats to the a ggregate amount of A llowed C laims of the same C lass plus all Disputed Claims in such Class.

1.96 "<u>Record Date</u>" means the record date for determining an entitlement to receive Distributions under the Plan on account of Allowed Claims, which with respect to holders of Allowed Claims (i) for the initial Distribution Date shall mean the Confirmation Date and (ii) for each subsequent Semi-Annual Distribution Date shall mean the twentieth (20) Business Day prior to such Semi-Annual Distribution Date.

"Reinstated" or "Reinstatement" means leaving unaltered the legal, 1.97 equitable, and contractual rights to which a Claim entitles the holder of such Claim so as to leave such Claim unimpaired in accordance with section 1124 of the Bankruptcy Code, thereby entitling the holder of such Claim to, but not more than, (a) reinstatement of the original maturity of the obligations on which such Claim is based, and (b) payment, as provided herein, of an amount of Cash consisting solely of the sum of (i) matured but unpaid principal installments, without regard to any acceleration of maturity, accruing prior to the Effective Date, (ii) accrued but unpaid interest as of the Petition Date, and (iii) reasonable fees, expenses, and charges, to the extent such fees, expenses, and charges are Allowed under the Bankruptcy Code and are specifically provided for in the agreement or agreements on which such Claim is based; provided, however, that any contractual right that does not pertain to the payment when due of principal and interest on the obligation on which such Claim is based, including, but not limited to, financial covenant ratios, negative pledge covenants, covenants or restrictions on merger or consolidation, and affirmative covenants regarding corporate existence or prohibiting certain transactions or actions contemplated by the Plan, or conditioning such transactions or actions on certain factors, shall not be required to be reinstated in order to accomplish Reinstatement.

1.98 "<u>Reorganized Dictaphone</u>" means, after the effective date of the Dictaphone Plan (i.e., March 28, 2002), Dictaphone and any successor thereby by merger, consolidation or otherwise.

1.99 "Schedules" means the schedules of assets and liabilities and the statements of financial affairs filed with the Bankruptcy Court on or about March 10, 2001 by the Debtor under section 521 of the Bankruptcy Code and the Official Bankruptcy Forms of the Bankruptcy Rules, as such schedules and statements existed on March 11, 2003, or may be amended, from time to time, with the prior written consent of the Committee, which consent may be withheld by the Committee in its sole and absolute discretion.

1.100 "<u>Semi-Annual Distribution Date</u>" shall have the meaning ascribed to such term in Section 7.2.1(b) of the Plan.

1.101 "Secured Claim" means a Claim secured by a Lien on Collateral to the extent of the value of the Collateral, as determined in accordance with section 506(a) of the Bankruptcy Code, or as otherwise agreed upon in writing by the Plan Administrator and subject to the approval of the Bankruptcy Court. To the extent that the value of such interest is less than the amount of the Claim which has the benefit of such security, the unsecured portion of such



Claim is an Unsecured Deficiency Claim unless, in any such case, the Class of which such Claim is a part makes a valid and timely election under section 1111(b) of the Bankruptcy Code to have such Claim treated as a Secured Claim to the extent Allowed.

1.102 "Securities Act" means the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder.

1.103 "Securities Class Actions/Suits" means the pending actions and lawsuits relating to the Securities Laws Claims.

1.104 "Securities Law Claim" means a Claim (1) arising from rescission of a purchase or sale of a security of the Debtor or an Affiliate of the Debtor; (2) for damages arising from the purchase or sale of such a security; (3) for reimbursement, indemnification, or contribution allowed under section 502 of the Bankruptey Code on account of a Claim for damages or rescission arising out of a purchase or sale of a security of the Debtor; or (4) for similar violations of the securities laws, misrepresentations, or any similar claim, including, to the extent related to the foregoing or subject to subordination under section 510(b) of the Bankruptey Code, but not limited to, any attorneys' fees, other charges or costs incorred in connection with the foregoing, claims for indemnification relating to the foregoing, the Securities Class Actions/Suita, and those certain Claims asserted by Stonington Capital Partners, Inc., Stonington Capital Appreciation 1994 Fund, L.P. and Stonington Holdings LLC.

1.105 "<u>Surplus Proceeds</u>" shall have the meaning ascribed to such term in Section 7.4.10(b) of the Plan.

1.106 "<u>Transfer</u>" means any sale, assignment, lease, transfer, encumbrance, Lien, exchange, mortgage, pledge, hypothecation or other disposition, or the creation of a security interest or encumbrance, in whole or in part.

1.107 "<u>United States Trustee</u>" means the United States Trustee appointed under section 591, title 28, United States Code to serve in the District of Delaware.

1.108 "<u>Unsecured Claim</u>" means any claim that is (i) a Bolgian Claim (ii) a Claim that is (a) the subject of a timely filed proof of claim, (b) listed in the Schedules (other than a claim listed as contingent, disputed or unliquidated), or (c) Allowed by a Final Order, and (iii) not an Administrative Expense Claim, Secured Claim, Priority Tax Claim, Priority Non-Tax Claim, PIERS/Old Convertible Subordinated Notes Claim, Belgian Priority Claim, or Securities Law Claim.

1.109 "Unsecured Deficiency Claim" means, with reference to a Claim secured by a Lien against Collateral, an amount equal to the difference between (a) the aggregate amount of such Claim after giving effect to the operation of section 1111(b)(1)(A) of the Bankruptcy Code and (b) the amount of such Claim that is a Secured Claim; provided, however, that, in the event that the Class in which such Secured Claim is classified makes the election under section 1111(b)(2) of the Bankruptcy Code in accordance with Rule 3014 of the Bankruptcy Rules, the



Unsecured Deficiency Claim otherwise relating to such Secured Claim shall be extinguished. An Unsecured Deficiency Claim is an Unsecured Claim.

1.110 "<u>Voting Deadline</u>" means the date set by the Baakruptcy Court pursuant to the Disclosure Statement Approval Order by which all Ballots for acceptance or rejection of the Plan must be received by the Committee or the Bankruptcy Court appointed balloting agent.

### SECTION II

### INTERPRETATION; APPLICATION OF DEFINITIONS, RULES OF CONSTRUCTION AND COMPUTATION OF TIME

Wherever from the context it appears appropriate, each term stated in either the singular or the plural shall include both the singular and the plural, and pronouns stated in the masculine, feminine or neuter gender shall include the masculine, feminine and neuter. For purposes of the Plan (a) any reference in the Plan to a contract, instrument, release, indenture, or other agreement or document being in a particular form or on particular terms and conditions means that such document shall be substantially in such form or substantially on such terms and conditions, (b) any reference in the Plan to an existing document or exhibit filed or to be filed means such document or exhibit as it may have been or may be amended, modified, or supplemented and (c) unless otherwise specified, all references in the Plan to Sections, Articles, and Exhibits are references to Sections, Articles, and Exhibits of or to the Plan. The words "herein," "hereof," "hereto," "hereunder," and other words of similar meaning refer to the Plan as a whole and not to any particular section, subsection or clause contained in the Plan. A capitalized term used herein that is not defined herein shall have the meaning assigned to that term in the Bankruptcy Code or in the exhibits hereto. The rules of construction contained in section 102 of the Bankruptcy Code shall apply to the construction of the Plan. The headings in the Plan are for convenience of reference only and shall not limit or otherwise affect the provisions of the Plan. Unless otherwise indicated herein, all references to dollars means United States dollars. In computing any period of time prescribed or allowed by the Plan, unless otherwise expressly provided, the provisions of Bankruptcy Rule 9006(a) shall apply.

### SECTION III

# PAYMENT OF ADMINISTRATIVE EXPENSE CLAIMS, PRIORITY TAX CLAIMS AND OTHER UNCLASSIFIED CLAIMS

3.1 <u>Administrative Expense Claims</u> Each holder of an Allowed Administrative Expense Claim shall be paid Cash in full by Post Effective Date L&H, (i) upon the Effective Date or as soon as practicable thereafter, (ii) as soon as practicable after such Claim becomes an Allowed Administrative Expense Claim if the date of allowance is later than the Effective Date, or (iii) upon such other terms as may be mutually agreed upon between such holder of an Allowed Administrative Expense Claim and Post Effective Date L&H; provided, however, that all post-Effective Date professional fees and related expenses accrued by Professionals in connection with the Plan and the Chapter 11 Assets shall be paid by Post Effective Date L&H within ten (10) Business Days of the submission by any Professional of an

invoice to Post Effective Date L&H. In the event that Post Effective Date L&H objects to the payment of a Professional's post-Effective Date invoice, in whole or part, and the parties cannot resolve such objection after good faith negotiation, the Bankruptcy Court shall retain jurisdiction to review the disputed invoice and make a determination as to the extent to which the invoice shall be paid by Post Effective Date L&H.

Priority Tax Claims On the Effective Date or as soon as practicable 3.2 thereafter, or as soon as practicable after such Claim becomes an Allowed Claim if the date of allowance is later than the Effective Date, a holder of an Allowed Priority Tax Claim shall be entitled to receive in full satisfaction, settlement, and release of, and in exchange for such Allowed Priority Tax Claim (a) deferred Cash payments in an aggregate principal amount equal to the amount of such Allowed Priority Tax Claim plus interest, to the extent required under applicable law, on the unpaid portion thereof at the legal rate of interest (excluding any default interest rate), or, in the absence of a legal rate of interest, at a rate of four percent (4%) per annum, from the Effective Date through the date of payment thereof, which date shall not extend beyond the sixth anniversary of the Effective Date, or (b) such other treatment as to which Post-Effective Date L&H and such holder shall have agreed upon in writing, with the approval of the Bankruptcy Court. If deferred Cash payments are made to a holder of an Allowed Priority Tax Claim, the remaining unpaid portion of such Allowed Claim shall be paid on or before the sixth anniversary of the Effective Date, together with any accrued and unpaid interest to the date of payment; provided, however, that Post Effective Date L&H reserves the right to pay any Allowed Priority Tax Claim, or any remaining balance of such Allowed Priority Tax Claim, in full at any time on or after the Effective Date without premium or penalty.

### SECTION IV

#### CLASSIFICATION OF CLAIMS AND EQUITY INTERESTS

4.1 <u>Summary</u> For purposes of all confirmation issues, including, without limitation, voting, confirmation and distribution, except as otherwise provided herein, all Claims against the Chapter 11 Assets (except for Administrative Expense Claims and Priority Tax Claims) and Equity Interests in the Debtor are classified as follows:



CLASS	CLASS NAME	STATUS
Class 1	Priority Non-Tax Claims	Unimpaired not entitled to vote
Class 2	Secured Claims	Unimpaired - not entitled to vote
Class 3	Unscented Claims	Impaired - entitled to vote
Class 4	PIERS/Old Convertible Subordinated Notes Claims	Impaired – entitled to vote
Class 5	Common Stock	Impaired – deemed to have rejected the plan and not entitled to vole
Class 6	Securities Law Claims	Impaired - deemed to have rejected the plan and not entitled to vote
Class 7	Other Equity Interests	Impaired deemed to have rejected the plan and not entitled to vote

### 4.1.1 Claims against and Equity Interests in L&H NV.

4.2 <u>Classification of Claims against the Chapter 11 Assets and Equity</u> Interests in L&H NV:

Non-Tax Claims.

4.2.1 <u>Class 1: Priority Non-Tax Claims</u>. Class 1 consists of all Priority

4.2.2 Class 2: Secured Claims. Class 2 consists of all Secured Claims.

4.2.3 <u>Class 3: Unsecured Claims</u>. Class 3 consists of all Unsecured Claims other than (a) Class 4 PIERS/Old Convertible Subordinated Notes Claims and (c) Class 6 Securities Laws Claims. Class 3 Unsecured Claims include, inter alia, Claims under the Belgian Revolving Credit Facility, all prepetition trade Claims relating to the Chapter 11 Assets, the Belgian Claims and other prepetition general unsecured Claims relating to the Chapter 11 Assets.

4.2.4 <u>Class 4: PIERS/Old Convertible Subordinated Notes Claims</u>. Class 4 consists of all PIERS Transaction Claims and Old Convertible Subordinated Notes Claims.

4.2.5 <u>Class 5: Common Stock</u>. Class 5 consists of all interests of the holders of Common Stock on account of such interests.

4.2.6 <u>Class 6: Scentities Law Claims</u>. Class 6 consists of all Securities

Law Claims.

4.2.7 <u>Class 7: Other Equity Interests</u>. Class 7 consists of all Equity Interests in L&H NV not otherwise classified in Classes 5 and 6, including the interests of holders of Old Stock Options.

#### SECTION V

#### TREATMENT OF CLAIMS AND EQUITY INTERESTS UNDER THE PLAN

5.1 <u>Designation of Treatment</u>. The following treatment set forth in this Section V shall be accorded to Claims against the Chapter 11 Assets and Equity Interests in the Debtor.

# 5.2 Claims Against the Chapter 11 Assets and Equity Interests in L&H NV.

5.2.1 <u>Class 1: Priority Non-Tax Claims</u>. On the Effective Date or as soon as practicable thereafter, or as soon as practicable after such Priority Non-Tax Claim becomes an Allowed Priority Non-Tax Claim if the date of allowance is later than the Effective Date, a holder of an Allowed Class 1 Priority Non-Tax Claim shall receive, in full satisfaction, settlement, release of, and in exchange for such Allowed Class 1 Priority Non-Tax Claim (a) a Distribution of Cash equal to the amount of such Allowed Class 1 Priority Non-Tax Claim, or (b) such other treatment as to which Post Effective Date L&H and such holder shall have agreed upon in writing.

5.2.2 <u>Class 2: Secured Claims</u>. Each holder of an Allowed Secured Claim shall be deemed to be classified in a separate Class and shall be treated as follows: to the extent that any such Claim is determined to be an Allowed, valid and perfected Secured Claim, the holder of such Secured Claim shall receive the first net proceeds (i.e., proceeds net of all costs and expenses related to such sale) from the sale of any of its Collateral to the extent of the principal amount of its Claim. To the extent permitted under applicable law, including the Bankruptcy Code, as determined by the Bankruptcy Court at the Confirmation Hearing, the holder of such Allowed Secured Claim shall receive the contractual non-default rate of interest on such Allowed Secured Claim semiannually in arrears based upon the amount of unpaid principal for such period and permitted costs thereon. Until each Secured Claim is paid in full, the holder of such Allowed Secured Claim shall retain the Liens securing such Allowed Secured Claim. Notwithstanding anything to the contrary herein, Post Effective Date L&H shall make fail p ayment to each such secured creditor to the extent of its Allowed Secured Claim on or before December 31, 2004.

#### 5.2.3 Class 3: Unsecured Claims.

(a) Treatment. Subject to Section 5.2.3(b) of the Plan, on the Effective Date or as soon as practicable thereafter, or if such Unsecured Claim becomes an Allowed Unsecured Claim after the Effective Date, as soon as practicable after such Unsecured Claim becomes an Allowed Unsecured Claim, and thereafter in full satisfaction, settlement, release of, and in exchange for such Allowed Class 3 Unsecured Claim, each holder of an Allowed Class 3 Unsecured Claim shall receive a Ratable Proportion of the Available Cash and ninety-three percent (93%) of the Litigation Trust Beneficial Interests (the "<u>Allowed Unsecured Claim Trust Interests</u>"); provided, however, that holders of Claims under the Belgian Revolving Credit Facility (which are classified under, and constitute a portion of, Class 3 Unsecured Claims) shalt not be entitled to any Distributions that are the proceeds (either directly or derivatively) of



distributions made to L&H NV pursuant to the Dictaphone Plan on a count of Intercompany Loan Agreement Claims assertable by L&H NV under the Dictaphone Plan; and <u>provided</u>, <u>further</u>, that as a condition to receiving the Distributions set forth in this Section 5.2.3 of the Plan, all holders of Claims under the Belgian Revolving Credit Facility shall be deemed to have waived any right to receive, and shall not receive, either directly or derivatively, any distributions that are made to L&H NV pursuant to the Dictaphone Plan on account of Intercompany Loan Agreement Claims. No interest shall be paid on any Class 3 Unsecured Claim.

(b) Limitations on Amount of Distributions to Holders of Allowed Unsecured Claims. Holders of Allowed Class 3 Unsecured Claims shall not receive any Distributions after they have received 100% repayment of the principal amount of their Allowed Class 3 Unsecured Claims. Any distributions by Dictaphone or Reorganized Dictaphone under the Dictaphone Plan, as the case may be, to the Lenders on account of the Dictaphone Guaranty shall be included in determining whether the Lenders have received 100% repayment of their claims relating to the Belgian Revolving Credit Facility. The Lenders, however, may assort the full amount of their Allowed Class 3 Unsecured Claims until such time as they have received 100% repayment of their facility (after taking into account distributions under the Plan and the Dictaphone Plan). After all holders of Allowed Class 3 Unsecured Claims have received 100% repayment of the principal amount of their Allowed Class 3 Unsecured Claims have received 100% repayment of the principal amount of their Allowed Class 3 Unsecured Claims have received 100% repayment of the principal amount of their Allowed Class 3 Unsecured Claims have received 100% repayment of the principal amount of their Allowed Class 4 PIERS/Old Convertible Subordinated Notes Claims.

5.2.4 <u>Class 4</u>: <u>PIERS/Old Convertible Subordinated Notes Claims</u>. Subject to Section 5.2.3(b) of the Plan, on the Effective Date or as soon as practicable thereafter, or, as soon as practicable after such Claim becomes an Allowed PIERS/Old Convertible Subordinated Notes Claim if the date of allowance is later than the Effective Date, and thereafter, each holder of a PIERS/Old Convertible Subordinated Notes Claim shall receive, in full satisfaction, settlement, release of, and in exchange for such Allowed PIERS/Old Convertible Subordinated Notes Claim, a Ratable Proportion of both (i) the Excess Available Cash (if any) and (ii) seven-percent (7%) of the Litigation Trust Beneficial Interests. No interest shall be paid on any Class 4 PiERS/Old Convertible Subordinated Notes Claim.

5.2.5 <u>Class 5: Common Stock</u>. Holders of Allowed Common Stock Equity Interests shall receive no Distributions under the Plan on account of such Class 5 Common Stock.

5.2.6 <u>Class 6</u>; <u>Securities Law Claims</u>. Holders of Allowed Class 6 Securities Law Claims shall receive no Distributions under the Plan on account of such Class 6 Securities Law Claims.

5.2.7 <u>Class 7: Other Equity Interests</u>. A holder of any Equity Interest in L&H NV not otherwise classified in Classes 5 or 6 shall receive no Distributions under the Plan on account of such Class 7 Other Equity Interest.



18

### SECTION VI

### IMPAIRED AND UNIMPAIRED CLASSES OF CLAIMS AND EQUITY INTERESTS UNDER THE PLAN; ACCEPTANCE OR REJECTION OF THE PLAN

# 6.1 Holders of Claims and Equity Interests Entitled to Vote.

(a) Each holder of an Allowed Claim, or the holder of a Claim that has been temporarily allowed for voting purposes only under Bankruptcy Rule 3018(a), in an impaired Class of Claims, shall be entitled to vote separately to accept or reject the Plan as provided in the Disclosure Statement Approval Order. Any unimpaired Class of Claims shall be deemed to have accepted the Plan. Any Class of Claims or Equity Interests that shall not receive or retain any property on account of such Claims or Equity Interests under the Plan shall be deemed to have rejected the Plan.

(b) Each of Classes 3 and 4 is impaired under the Plan, and the holders of Allowed Claims in such Classes are entitled to vote on the Plan. In accordance with section 1126(g) of the Bankruptcy Code, each of Classes 5, 6, and 7 is conclusively deemed to have rejected the Plan.

6.2 <u>Acceptance by Unimpaired Classes</u>. Each of Classes 1 and 2 is unimpaired under the Plan and each such Class is conclusively presumed to have accepted the Plan pursuant to section 1126(f) of the Bankruptcy Code.

6.3 <u>Elimination of Vacant Classes</u>. Any Class of Claims that is not occupied as of the date of the commencement of the Confirmation Hearing by an Allowed Claim or a Claim temporarily allowed under Bankruptcy Rule 3018 shall be deemed deleted from the Plan for purposes of voting on or rejection of the Plan, and for purposes of determining acceptance or rejection of the Plan by such Class under section 1129(a)(8) of the Bankruptcy Code.

6.4 <u>Non-consensual Confirmation</u>. The Committee shall request that the Bankruptcy Court confirm the Plan in accordance with section 1129(b) of the Bankruptcy Code in view of the deemed rejection of the Plan by Classes 5, 6, and 7. In the event that any of Classes 3 and 4 fail to accept the Plan, the Committee reserves the right (i) to modify the Plan in accordance with Section 14.2 hereof and/or (ii) to request that the Bankruptcy Court confirm the Plan in accordance with section 1129(b) of the Bankruptcy Code notwithstanding such lack of acceptance by finding that the Plan provides fair and equitable treatment to, and does not discriminate unfairly against, any impaired Class of Claims and Equity Interests voting to reject the Plan.

 $6.5 = \frac{\text{Revocation of the P lan}}{1.5}$ . Subject to Section 15.8 hereof, the Committee may revoke and withdraw the Plan at any time prior to ontry of the Confirmation Order. If the Plan is so revoked or withdrawn, then it shall be deemed null and void.



#### SECTION VII

### MEANS OF IMPLEMENTATION OF THE PLAN

7.1 <u>Transactions on the Effective Date</u>. On the Effective Date, the following shall occur (and shall be deemed to have occurred simultaneously):

- (a) the Litigation Trust Agreement shall become effective;
- (b) the Litigation Trust shall be formed and shall assume its obligations hereunder;
- (c) the Plan Administration Agreement shall become effective; and
- (d) the Equity Interests (including the Old Capital Stock) of L&H NV shall be extinguished as they relate to the Chapter 11 Assets.

in no event shall any of the foregoing events occur unless all of the foregoing events occur on the same Business Day.

7.2 <u>Approval of Settlements</u>. The entry of the Confirmation Order shall constitute the Bankruptcy Court's finding and determination that the settlements reflected in the Plan, including, but not limited to, (i) the settlement of the Distribution allocations between the various Classes of creditors and (ii) the allocation of assets between the Belgian Case and the Chapter 11 Case, are (a) in the best interest of L&H NV and its Estate, (b) fair, equitable and reasonable, (c) made in good faith, and (d) approved by the Bankruptcy Court. Subject to obtaining the approval of the settlements reflected in this Plan by the Bankruptcy Court, on the Effective Date, Post Effective Date L&H shall take all actions necessary or reasonably required to effectuate the matters set forth in such settlements.

7.2.1 Periodic Distributions of Available Cash and Litigation Trust Beneficial Interests. Post Effective Date L&H shall make Distributions as provided under the Plan from the net proceeds it has obtained or obtains from the Transfers of the Chapter 11 Assets for a period of up to five (5) years after the Effective Date. To effect the foregoing, Post Effective Date L&H shall make Distributions of Cash, Available Cash, and Litigation Trust Beneficial Interests on account of any portion of, or in the full amount of Allowed Claims as soon as reasonably practicable after the Effective Date and, thereafter, on the first Business Day of each calendar semester (each such date, a "Semi-Annual Distribution Date"); provided, however, that, except with respect to the final Distribution, Post Effective Date L&H shall not make any such Distributions unless the amount of Available Cash is in excess of \$1,000,000. Such Distributions shall continue until Post Effective Date L&H has Transferred all of its assets and there is no additional Available Cash for Distributions under the Plan. To the extent that the Chapter 11 Assets consist of neither Cash nor Cash equivalents, Post Effective Date L&H shall endeavor to distribute such non-Cash consideration in such manner as to give effect to the distribution scheme contemplated under the Plan. Post Effective Date L&H shall have absolute discretion to pursue or not to pursue any and all claims, rights, defenses, or Causes of Action that



20

it retains pursuant to the Plan, as it determines in the exercise of its business judgment, and shall have no liability for the outcome of its decision.

7.3 Post Effective Date L&H. On the Effective Date, or as soon as reasonably practicable after the Effective Date, the Debtor shall Transfer all of the Chapter 11 Assets to Post Effective Date L&H. From and after the Effective Date, as set forth herein, the Plan Administrator, Post Effective Date L&H, the Litigation Trustee, and the Litigation Trust, as the case may be, shall perform their respective obligations under the Plan. The Plan shall be administered and actions shall be taken in the name of the Debtor and Post Effective Date L&H through, in accordance with the terms hereof, Post Effective Date L&H, the Plan Administrator, the Litigation Trustee, and/or the Litigation Trust, irrespective of whether the Debtor is dissolved. From and after the Effective Date, Post Effective Date L&H shall continue in existence for the purpose of:

- (a) administering the Plan and to take all steps and execute all instruments and documents necessary to effectuate the Plan;
- (b) selling or otherwise disposing of the Chapter 11 Assets and winding up affairs relating to the Chapter 11 Assets as expeditiously as reasonably possible;
- (c) taking any actions to liquidate, and maximize the value of, the Chapter 11 Assets;
- (d) assigning all Assigned Causes of Action, and all claims, interests, rights and privileges of L&H NV relating thereto to the Litigation Trust for enforcement, prosecution and settlement by the Litigation Trustee in accordance with the terms of this Plan and the Litigation Trust Agreement; provided, however, that Post Effective Date L&H shall retain an interest in the Assigned Causes of Action solely to assert a defense to a Claim or Equity Interest based upon such Assigned Causes of Action;
- (c) reconciling Claims and resolving Disputed Claims, and administering the Claims allowance and disallowance processes as set forth in the Plan, including objecting, prosecuting, litigating, reconciling, settling and resolving Claims and Disputed Claims in accordance with the Plan;
- (f) making decisions regarding the retention, engagement, payment and replacement of professionals, employees and consultants;
- (g) cooperating with the Litigation Trustee, the Litigation Trust, and the Litigation Monitoring Committee with regard to the pursuit of the Assigned Causes of Action;

- (h) in conjunction with Litigation Trustee, providing quarterly reports to the Litigation Monitoring Committee as to budgets, cash receipts and disbursements, asset sales or other dispositions, claims reconciliation, Litigation Proceeds and Distributions under the Plan;
- (i) administering the Distributions under the Plan, including (i) making Distributions in accordance with the terms of the Plan, (ii) establishing and maintaining the various Disputed Claims Reserves, and (iii) filing with the Bankruptcy Court semi-annual reports regarding the Distributions to be made to the holders of Allowed Claims;
- (j) excreising such other powers as necessary or prudent to carry out the provisions of the Plan;
- (k) investing any Cash in any reserves or pending distribution in accordance with reasonable business judgment for any such Entity;
- (1) filing appropriate tax returns; and
- (m) taking such other action as may be necessary or appropriate to effectuate this Plan.

Each of the Plan Administrator, Post Effective Date L&H, the Litigation Trust, and the Litigation Trustee may incur and pay any reasonable and necessary expenses in performing the foregoing functions, subject to the terms of the Plan. For purposes of exercising its powers, each of the Plan Administrator, P ost E flective D ate L&H, the Litigation Trustee and the Litigation Trust shall be deemed to be a representative of the E state pursuant to section 1 123(b)(3)(B) of the Bankruptcy Code.

# 7.4 <u>The Litigation Trust.</u>

7.4.1 <u>The Establishment of the Litigation Trust</u>. The Litigation Trust shall be established as of the Effective Date. The Litigation Trustee shall be appointed on the Effective Date. The Litigation Trust will be established for the sole purpose of liquidating its assets, in accordance with Treasury Regulation Section 301.7701-4(d), with no objective to continue or engage in the conduct of a trade or business. Subject to definitive guidance from the Internal Revenue Service, all parties will treat the Litigation Trust as a liquidating trust for all federal income tax purposes.

7.4.2 <u>Acquisition of Assigned Causes of Action</u>. On the Effective Date, except as otherwise provided in the Plan, Post Effective Date L&H and the Debtor shall be deemed to have, and shall have, irrevocably assigned and transferred to the Litigation Trust all of their rights, title and interest in and to any and all of the Assigned Causes of Action and any proceeds thereof received by the Debtor or Post Effective Date L&H. Each of the Assigned Causes of Action, except as otherwise provided in the Plan, shall be free and clear of all Liens,



claims, encombrances and other interests. Except as otherwise provided in the Plan, neither Post Effective Date L&H nor the Debtor shall have any further right, title or interest in any of the Assigned Causes of Action, and neither the Debtor nor Post Effective Date L&H shall be entitled to receive any portion of any amounts recovered on account of any of the Assigned Causes of Action. Subject to the Curators and the Litigation Trustee entering into an agreement, if necessary, preserving confidential information and all applicable privileges, the Litigation Trustee shall consult with and make information available to the Curators with respect to: (a) any Causes of Action of L&H NV against former directors, officers, administrators or administrators in fact of L&H NV, arising pursuant to Clause 530 of the "Wetboek Van Venneotschappen" (the Belgian Company Code) dated May 7, 1999; (b) any and all Causes of Action arising out of or relating to any criminal proceeding initiated against former directors, officers, administrators or administrators in fact of L&H NV, including but not limited to, Causes of Action for indemnification; and (c) any and all Causes of Action the enforcement, prosecution, litigation or settlement of which either must be initiated, or already have been initiated, outside of the jurisdiction of the Courts of the United States.

7.4.3 <u>Establishment of Litigation Trust Beneficial Interest Registers</u>. The Litigation Trust shall maintain a register of the persons or entities granted Litigation Trust Beneficial Interests therein and shall be entitled to treat as the owner of any such interest for all purposes the person or entity in whose name the Litigation Trust Beneficial Interest is registered. Litigation Trust Beneficial Interests shall be uncertified.

7.4.4 <u>Litigation Trust Beneficial Interests Granted on Account of</u> <u>Disputed Claims</u>. Upon a Disputed Claim becoming an Allowed Claim, Post Effective Date L&H shall notify the Litigation Trust, and the holder of such Allowed Claim shall be granted the Litigation Trust Beneficial Interests reserved for such Claim.

7.4.5 <u>Limitations on Transferability of Litigation Trust Beneficial</u> Interests. No holder of a Litigation Trust Beneficial Interest shall be entitled to transfer such interest.

7.4.6 <u>Transfer of Causes of Action by Holders</u>. The irrevocable transfer of the Assigned Causes of Action to the Litigation Trust shall be treated as an irrevocable deemed transfer of such Assigned Causes of Action to the holders of Allowed Claims followed by an irrevocable deemed contribution of such Assigned Causes of Action by such holders of Allowed Claims to the Litigation Trust.

7.4.7 <u>Funding the Litigation Trust</u>. On the Effective Date, Post Effective Date L&H shall pay to the Litigation Trust the aggregate amount of \$1,000,000 for the establishment of a reserve to pay the fces, expenses and costs of the Litigation Trust and the Litigation Trustee (such reserve, the "Litigation Trust Reserve"). In addition, on the Effective Date, Post Effective Date L&H shall pay to the Litigation Trust the aggregate amount of \$100,000 for the catablishment of a reserve to pay the fees, expenses and costs of the Litigation of \$100,000 for the catablishment of a reserve to pay the fees, expenses and costs of the Litigation Monitoring Committee (such reserve, the "Litigation Monitoring Committee Reserve"). To the extent that the Litigation Trustee and the Litigation Monitoring Committee from time to time reasonably agree that the amounts of the Litigation Trust Reserve and/or the Litigation



Monitoring Committee Reserve are in excess of the amounts reasonably anticipated to be incurred by the Litigation Trust and the Litigation Trustee in the pursuit of their respective duties and obligations hereunder, such excess amounts shall be returned to the Plan Administrator and shall be treated as Available Cash for Distributions to holders of Allowed Claims. To the extent that the Litigation Monitoring Committee reasonably determines that the amount of the Litigation Trust Reserve is not sufficient for the Litigation Trustee to pursue its duties and obligations hereunder, the Litigation Monitoring Committee may request from Post-Effective Date L&H additional funding for the Litigation Trust Reserve, and the consent of Post-Effective Date L&H with respect to such request for additional funding for the Litigation Trust Reserve shall not be withheld unreasonably. Except as specifically set forth in the Plan and the Litigation Trust Agreement, Post Effective Date L&H shall have no obligation to the Litigation Trust or any holder of an interest therein other than its obligations to reasonably cooperate as a party to the Assigned Causes of Action assigned to the Litigation Trust and to fund the Litigation Trust and the Litigation Monitoring Committee Reserves. In addition, the Litigation Trustee may, with the written consent of the Litigation Monitoring Committee, borrow funds to finance the operations of the Litigation Trust, which borrowing(s) may include equity participation features.

Upon receipt of the 7,4.8 Application of Proceeds and Expenses. proceeds of any Assigned Causes of Action assigned to the Litigation Trust, the Litigation Trustee shall determine whether to distribute such proceeds to holders of Litigation Trust Beneficial Interests. Prior to any Distribution, the Litigation Trustee shall apply such proceeds. net of amounts paid or deductions made by reason of set-off to the Litigation Trustee or by reason of reduction in judgment or reimbursement obligations of the Litigation Trustee, as follows: (i) first, after utilizing amounts in the Litigation Trust Reserve and the Litigation Monitoring Committee Reserve, to the payment of any associated taxes and unpaid administrative expenses of the Litigation Trust, the Litigation Trustee and the Litigation Monitoring Committee, (ii) second, pro rata to the payment of the reasonable unpaid fees and expenses incurred in employing professionals for the Litigation Trustee and the Litigation Monitoring Committee, and the compensation and expenses of the Litigation Trustee, (iii) third, to either the Litigation Trust Reserve or the Litigation Monitoring Committee Reserve for the reasonably anticipated amount of any future expenses and obligations to the extent the amounts in such reserves are insufficient and (iv) fourth, 23.3% of amounts remaining after the payment of items i-iii above shall be distributed to the Curators for the benefit of priority creditors in the Belgian Case, provided, however, that the aggregate amount of all such payments shall not exceed  $\in$  2 million (collectively, (i)-(iv), the "Litigation Administrative Costs"). Notwithstanding anything contained herein to the contrary, and for purposes of this Section 7.4.8 only, the conversion rate a mong U nited States Dollars (\$) and the E uro ( $\mathfrak{E}$ ) shall be the spot conversion rate expressed in Dollars per Euro published at 10:00 a.m. (EST) by the Federal Reserve Bank of New York on the Effective Date.

7.4.9 <u>Distribution of Litigation Proceeds</u>. Subject to the limitations set forth in Section 7.4.10(b), any proceeds from the Assigned Causes of Action distributed by the Litigation Trustee, net of amounts paid or reductions made by reason of setoff and after payment or reserve in full of the Litigation Administrative Costs of the Litigation Trust (such net proceeds, the "Litigation Proceeds"), shall be distributed in accordance with Section 7.4.10 based on each holder's Ratable Proportion. Except as otherwise provided in this Section 7.4 of the



Plan, Distributions of the Litigation Proceeds shall be governed by Sections V and VIII of the Plan.

#### 7.4.10 Distribution by Litigation Trustee.

(a) <u>Timing of Distributions</u>. The Litigation Trustee, shall distribute the Litigation Proceeds at such times as the Litigation Trustee deems appropriate, but only after, pursuant to this Section 7.4 of the Plan, paying all outstanding Litigation Administrative Costs and reserving for any additional reasonable Litigation Administrative Costs that may be incurred thereafter.

(b) <u>Maximum Recovery</u>. Until all holders of Allowed Claims receive full payment of their Allowed Claims, the aggregate Distributions on account of any holder of an Allowed Claim shall not exceed 100% of the amount of such Allowed Claim (such amount, the "<u>Maximum Recovery Amount</u>"). Any Litigation Proceeds otherwise distributable to a holder of an Allowed Claim in excess of the Maximum Recovery Amount (the "<u>Sumplus Proceeds</u>") shall be redistributed to holders holding the same class of Claims or, if all such holders have received 100% of the amount of such Allowed Claims, to holders of Allowed Claims in proportion to the allocation of Distributions between the various Classes of Claims against the Chapter 11 Assets as set forth in Section V of the Plan.

7.4.11 <u>Litigation Reserve</u>. The Litigation Trust shall withhold from the amounts to be distributed to holders of Litigation Trust Beneficial Interests amounts sufficient to be distributed on account of the Litigation Trust Beneficial Interests that may be granted to holders of Claims that are Disputed Claims as of the date of distribution of proceeds, and the Litigation Trust shall place such withhold proceeds in reserve (the "Litigation Reserve"). To the extent such Disputed Claims ultimately become Allowed Claims, and Litigation Trust Beneficial Interests are granted to the holders of such Claims in accordance with Section 7.4.4 of the Plan, payments with respect to such interest shall be made from the Litigation Reserve. The Litigation Trustee, in consultation with the Litigation Monitoring Committee, shall determine the amount to reserve in the Litigation Reserve based on the amount of Disputed Claims determined or estimated for the purposes of the Disputed Claims Reserve.

7.4.12 <u>Distributions After Disallowance</u>. If any proceeds remain in the Litigation Reserve after all objections to Disputed Claims have been resolved, such remaining amounts shall be distributed as soon as practicable in accordance with the provisions regarding the distribution of Litigation Proceeds to the holders of Litigation Trust Beneficial Interests.

7.4.13 <u>Term</u>. The term of the Litigation Trust shall commence on the Effective D ate and shall continue until the fifth (5th) a universary of such date, unless sooner terminated in accordance with the terms of the Litigation Trust Agreement; provided, however, that the term of the Litigation Trust may be extended for no more than five (5) successive periods of two years each in accordance with the provisions in the Litigation Trust Agreement.

7.4.14 <u>Powers and Duties of the Litigation Trustee</u>. As more fully described in the Litigation Trust Agreement, with respect to the Litigation Trust, the Litigation Trustee shall at all times act solely for the benefit of the holders of Litigation Trust Beneficial



Interests and shall not act for the benefit of or have any obligations to or on behalf of the Debtor or Post Effective Date L&H. In addition to those other powers and duties set forth in the Litigation Trust Agreement, the Litigation Trustee shall be empowered: (a) to take all steps and execute all instruments and documents necessary to effectuate the Litigation Trust, (b) to pay all Litigation Administrative Costs of the Litigation Trust, (c) to comply with the Plan and the Litigation Trust Agreement and the obligations thereunder, (d) to employ, retain or replace professionals to represent it with respect to its responsibilities, (e) to waive or enforce, to the fullest extent permitted by law, any existing client privilege of the Debtor, (f) to investigate any claims, rights, Assigned Causes of Action or other Causes of Action assigned to the Litigation Trust, (g) to prosecute, litigate, settle, adjust, retain, enforce or abandon any claims, rights, Assigned Causes of Action or other Causes of Action assigned to the Litigation Trust, including any counterclaims to the extent such counterclaims are setoff against the proceeds of any such Causes of Action, (h) to exercise such other powers as may be vested in the Litigation Trustee pursuant to an order of the Bankruptcy Court or the Liligation Trust Agreement, or as deemed by the Litigation Trustee to be necessary and proper to carry out the provisions of the Litigation Trust, and (i) to make distributions contemplated by the Litigation Trust to holders of Litigation Trust Beneficial Interests. Subject to the terms of Section 7.4.15 of the Plan, and to the requirement to consult with the Litigation Monitoring Committee, the Litigation Trustee shall have sole and absolute discretion to hold, pursue, prosecute, release, settle or abandon, as the case may be, any and all claims, rights, Assigned Causes of Action or other Causes of Action assigned to the Litigation Trust pursuant to this Plan, as it determines in the exercise of its business judgment, and shall have no liability for the outcome of its decisions. For purposes of exercising its powers, each of the Litigation Trustee and the Litigation Trust shall be deemed to be a representative of the Debtor pursuant to section 1123(b)(3)(B) of the Bankruptcy Code. To the extent that the Litigation Trustee abandons any Assigned Causes of Action, such Assigned Causes of Action shall revest and be deemed to be held by Post Effective Date L&H without any further act of the Bankruptcy Court or any other party.

7.4.15 Authority to Settle Causes of Action. With respect to any claims, rights, Assigned Causes of Action, or other Causes of Action assigned to the Litigation Trust in which the asserted amount is equal to or less than \$200,000, the Litigation Trustee shall be empowered and authorized, without approval of the Bankruptcy Court or notice to any other Person, to settle, adjust, dispose of or abandon any such claims, rights, Assigned Causes of Action or other Causes of Action assigned to the Litigation Trust, including any counterclaims to the extent such counterclaims are setoff against the proceeds of any such Causes of Action, upon written notice to the Litigation Monitoring Committee. If the Litigation Monitoring Committee disputes the reasonableness of the proposed settlement, disposition or abandonment, on or prior to the conclusion of ten (10) Business Days after the Litigation Monitoring Committee receives written notice of a proposed settlement, disposition or abandonment, or any extension thereof agreed upon between the Litigation Trustee and the Litigation Monitoring Committee, the Litigation Monitoring Committee can submit to the Bankruptcy Court for resolution the reasonableness of the Litigation Trustee's proposed settlement, disposition or abandonment. With respect to any claims, rights, Assigned Causes of Action, or other Causes of Action assigned to the Litigation Trust, in which the asserted amount is greater than \$200,000, the Litigation Trustee shall (i) obtain the written consent of the Litigation Monitoring Committee and (ii) obtain approval of the Bankrapicy Court to settle, adjust, dispose of or abandon any such

26

claims, rights, Assigned Causes of Action or other Causes of Action in accordance with Bankniptcy Rule 9019.

7.4.16 <u>Compensation of the Litigation Trustee</u>. The Litigation Trustee shall receive compensation for services to the Litigation Trust as agreed upon by the (a) Litigation Trustee and (b) the Litigation Monitoring Committee, for its services and shall be entitled to reimbursement of reasonable expenses incurred in performing its duties hereunder out of the assets of the Litigation Trust, as more fully set forth in the Litigation Trust Agreement or in a separate agreement setting forth the terms and conditions of the provision of such services by the Litigation Trustee.

7.4.17 Indemnification and Exculpation of the Litigation Trustee and Litigation Monitoring Committee.

(a) From and after the Effective Date, the Litigation Trustee and members of the Litigation Monitoring Committee, and each of their respective post-Effective Date directors, members, officers, shall be exculpated and indemnified as provided in the Litigation Trust Agreement.

Except as may otherwise be provided in the Litigation Trust Agreement, (b) both Litigation Trustee and the Litigation Monitoring Committee, and each of their post-Effective Date directors, members, officers, employees, agents and attorneys, as the case may be, from and after the Effective Date, is hereby exculpated by all Persons, holders of Claims and Equity Interests, Entities, and parties in interest receiving Distributions under the Plan, from any and all claims, causes of action, and other assertions of liability arising out of its discharge of the powers and duties conferred upon it by the Plan, the Litigation Trust Agreement, or any order of the Bankruptcy Court entered pursuant to or in furtherance of the Plan, or applicable law, except solely for actions or emissions arising out of its gross negligence, willful misconduct or breach of its fiduciary duties. No holder of a Claim or an Equity Interest, or representative thereof, shall have or pursue any claim or cause of action (a) against the Litigation Trustee or the Litigation Monitoring Committee (or their respective directors, members, officers, employees, agents and attomeys, as the case may be) for Distributions made in accordance with the Plan, or for implementing the provisions of the Plan, or (b) against any holder of a Claim for receiving or retaining payments or other Distributions as provided for by the Plan.

(c) Both the Litigation Trustee and the Litigation Monitoring Committee shall not be liable for any action taken or omitted in good faith and reasonably believed by it to be authorized within the discretion or rights or powers conferred upon it by the Plan or the Litigation Trust A greement. In performing its duties under the Plan and the Litigation Trust Agreement hereunder, each of the Litigation Trustee and the Litigation Monitoring Committee shall have no liability for any action taken by the Litigation Truster and the Litigation Monitoring Committee in good faith in accordance with the advice of counsel, accountants, appraisers and other professionals retained by it. Post Effective Date L&H, or the Litigation Trust. Without limiting the generality of the foregoing, the Litigation Trustee and the Litigation Monitoring Committee may rely without independent investigation on copies of orders of the Bankruptcy Court reasonably believed by it to be genuine, and shall have no liability for actions


taken in good faith in reliance thereon. None of the provisions of the Plan shall require the Litigation Trustee or the Litigation Monitoring Committee to expend or risk their own funds or otherwise incur personal financial liability in the performance of any of their duties hereonder or in the exercise of any of its rights and powers. Each of the Litigation Trustee and the Litigation Monitoring Committee may rely without inquiry upon writings delivered to it hereunder which it reasonably believes in good faith to be genuine and to have been given by a proper Person.

# 7.4.18 Monitoring of Litigation Trust.

(a) <u>Appointment: Responsibilities</u>. On or prior to the Confirmation Date, the Committee shall select three (3) Persons or Entities to serve as representatives of the Litigation Trust, (collectively, the "<u>Litigation Monitoring Committee</u>"). A list setting forth the identities of the members of the Litigation Monitoring Committee, to the extent available, shall be filed as part of the Plan Supplement, or otherwise, in a submission to this Bankruptey Court on or prior to the Effective Date. The Litigation Monitoring Committee shall monitor and review settlement, abandonment and other disposition proposals proposed to or agreed to by the Litigation Trustee with respect to the Assigned Causes of Action and to consult with the Litigation Trustee regarding the settlement, abandonment, disposition and prosecution of such Causes of Action. The Litigation Monitoring Committee shall, to the extent it deems necessary, rotain counsel to assist it.

(b) <u>Quorum: Meetings: Votes.</u> A quorum for the Litigation Monitoring Committee shall consist of a majority of the then existing members of the Litigation Monitoring Committee. No meeting of the Litigation Monitoring Committee shall be held unless a quorum is present at the beginning of any meeting. Any member of the Litigation Monitoring Committee may call for a meeting to be convened upon notice of such meeting being given at least two (2) Business Days prior to the proposed date of the meeting, which notice may be given by telephone, overnight mail, or facsimile transmission, to each of the other members of the Litigation Monitoring Committee. Meetings shall be held in person or by telephone conference call. Any action by the Litigation Monitoring Committee shall require the affirmative vote of a majority of those voting provided that a quorum is present at the time of the vote.

(c) <u>Resignations: Successor Members</u>. Any Entity may resign as a member of the Litigation Monitoring Committee at any time. In the event that any Entity resigns as a member of the Litigation Monitoring Committee, the remaining members of the Litigation Monitoring Committee shall select a replacement for such Entity; <u>provided</u>, <u>however</u>, that any such replacement must hold Litigation Trust Beneficial Interests.

(d) <u>Removal of Litigation Trustee</u>. Upon a unanimous determination by the members of the Litigation Monitoring Committee, the Litigation Trustee may be removed by the Litigation Monitoring Committee without any necessity for any showing of cause. In addition, (a) a majority vote of the Litigation Monitoring Committee and (b) a vote among twenty-five percent (25%) of the beneficiaries of the Litigation Trust can seek to have the Litigation Trustee removed for cause. To the extent there is any dispute regarding the removal of a Litigation Trustee, the Bankruptcy Court shall retain jurisdiction to consider and adjudicate any such dispute and may remove a Litigation Trustee for cause.



7.4.19 <u>Discretion Afforded to Litigation Monitoring Committee to</u> <u>Modify Terms of Compensation of Litigation Trustee</u>. Notwithstanding anything to the contrary in the Plan, the Litigation Monitoring Committee may, with the unanimous consent of each of the members of the Litigation Monitoring Committee, and without application to or approval by the Bankruptey Court, subject to the consent of the Litigation Trustee, modify the Litigation Trustee's compensation and other terms regarding the retention of the Litigation Trustee.

7.5 <u>Cancellation of Equity Interests</u>. Except to the extent specifically provided otherwise in the Plan, on the Effective Date, all existing Equity Interests shall, without any further action, be cancelled, annulled and extinguished and any certificates representing such Equity Interests shall be null and void with respect to the Chapter 11 Assets.

7.6 <u>Operations of L&H NV Between Confirmation and the Effective Date</u>. L&H NV shall continue to operate as a Debtor In Possession during the period from the Confirmation Date through and until the Effective Date.

7.7 <u>Closing of Chapter 11 Case</u>. When all Disputed Claims filed against the Chapter 11 Assets become Allowed Claims or have become a Disallowed Claim, and all remaining assets have been liquidated and converted into Cash, and such Cash has been distributed in accordance with the terms of this Plan, or at such earlier time as the Plan Administrator decins appropriate, the Plan Administrator shall seek authority from the Bankruptcy Court to close the Chapter 11 Case in accordance with the Bankruptcy Code and the Bankruptcy Rules.

7.8 <u>Exclusivity Period</u>. Subject to further order of the Bankreptoy Court, the Committee shall retain the exclusive right to amend the Plan and solicit acceptances thereof until the Effective Date (or until the earliest date on which the Effective Date can no longer occur pursuant to Section 11.4 of the Plan).

7.9 <u>Revesting of Assets</u>. The Chapter 11 Assets shall vest in Post Effective Date L&H on the Effective Date. From and after the Effective Date, Post Effective Date L&H may use, acquire, and dispose of property free of any restrictions imposed under the Bankruptcy Code, the Bankruptcy R ules, and the Bankruptcy Court. A softhe Effective Date, all of the Chapter 11 Assets and Post Effective Date L&H shall be free and clear of all Claims, Liens and interests, except as specifically provided in the Plan or in the Confirmation Order. Without limiting the foregoing, Post Effective Date L&H may, without application to or approval by the Bankruptcy Court, pay Professional Fees and expenses that Post Effective Date L&H may incur after the Effective Date.

7.10 <u>Committee</u>. The Committee (a) shall cease to exist on the Effective Date, provided that the Committee shall retain standing to appear at any hearing regarding the allowance of Professional Fees, (b) as appropriate, may interpose objections to such Professional Fees, and (c) shall be entitled to obtain reimbursement for the reasonable fees and expenses of its professionals relating to the foregoing.

7.11 Effectuating Documents; Further Transactions. Post Effective Date L&H shall be, and hereby is, authorized to execute, deliver, file, and record such contracts,

instruments, releases, indentures, certificates, and other agreements or documents, and take such other actions as may be necessary or appropriate to effectuate and further evidence the terms and conditions of the Plan.

7.12 <u>Assumptions of Liabilities</u>. The liabilities and obligations to make the Distributions as required under Sections III, IV and V of the Plan shall be assumed by Post Effective Date L&H, which shall have the liability for, and obligation to make, all Distributions of Cash, Available Cash, the Litigation Trust Beneficial Interests, or any other consideration or securities to be issued or distributed by Post Effective Date L&H under the Plan.

7.13 <u>Substantial Consummation</u>. Substantial consummation of the Plan ender section 1101(2) of the Bankruptcy Code shall not be deemed to occur, the Chapter 11 Case shall remain open and not be deemed fully administered, and no final decree closing the Chapter 11 Case shall be entered pursuant to section 350(a) of the Bankruptcy Code and Bankruptcy Rule 3022, until the Effective Date.

7.14 <u>Preservation of Certain Causes of Action</u>. Except as otherwise provided in the Plan, including in Section 13.1 and Section 13.2 of the Plan, in accordance with section 1123(b) of the Bankruptcy Code, Post Effective Date L&H, as successor-in-interest to the Debtor, shall retain and may enforce all of the Debtor's claims, rights, and Causes of Action that are the property of the Estate (including Avoidance Actions (except to the extent such Avoidance Actions constitute Assigned Causes of Action or except as provided under Section 13.1 and Section 13.2 of the Plan)), and Post Effective Date L&H shall retain and enforce all defenses and counterclaims to all Claims asserted against the Chapter 11 Assets, including, but not limited to, setoff, recomparent, and any rights under section 502(d) of the Bankruptcy Code. Post Effective Date L&H may pursue such claims, rights, Causes of Action, counterclaims, and defenses, as appropriate, in accordance with its best interests, as determined by Post Effective Date L&H.

7.15 <u>Cancellation of Existing Securities</u>. On the Effective Date, except as otherwise provided for herein, (i) all securities, equity interests, notes, bonds, indentures, and other instruments or documents evidencing or creating any indebtedness. Equity Interest or obligation of L&H NV shall be as against the Estate and its respective successors extinguished and canceled, and (ii) the obligations of L&H NV under any agreements, indentures, or certificates of designation governing any securities, equity interests, notes, bonds, indentures, and other instruments or documents evidencing or creating any indebtedness, Equity Interest or obligation of the Debtor relating to the Chapter 11 Assets, as the case may be, shall be discharged. Notwithstanding anything to the contrary set forth herein, the extinguishment or cancellation of notes, bonds, or indentures shall not alter, impair, or affect any rights of holders of notes or bonds to receive any and all distributions in accordance with the terms of the Plan, or any rights of the PIERS Indenture Trustee or the Old Convertible Subordinated Notes Indenture Trustee to receive any amounts due and payable under the terms of the Plan, including, but not limited to, Indenture Trustees' Fees, all of which shall survive confirmation of the Plan.

7.16 <u>Appointment of Scott L. Baena to Administer Plan</u>. On the Effective Date, Scott L. Baena shall be appointed to serve as the sole officer and director of Post Effective Date L&H pursuant to the Plan Administration Agreement. Sufficient funds shall be retained by

30

Post Effective Date L&H through the Final Distribution Date to fund the retention of Scott L. Baena and the performance of the duties of Scott L. Baena and Post Effective Date L&H under the Plan. Scott L. Baena shall be exclusively responsible for making all Distributions of Cash and Litigation Trust Beneficial Interests under the Plan.

7.16.1 <u>Distributions Under Plan</u>. On the Effective Date or as soon as practicable thereafter, or on such later date that Distributions are required to be made on account of Allowed Claims, Post Effective Date L&H shall make, or shall make adequate reserve for, the Distributions required to be made to all holders of Claims (whether or not Allowed) under the Plan. Cash necessary to make the Distributions required under the Plan shall be provided from all excess Cash of Post Effective Date L&H (if any), or any other source. All Distributions reserved pursuant to this Section shall be held by Post Effective Date L&H, for the benefit of the holders of Claims entitled to receive such Distributions. The Plan Administrator shall place Cash Distributions reserved under the Plan in one or more interest bearing accounts in the United States, as the Plan Administrator determines may be necessary or appropriate to effectuate the provisions of the Plan, and such Cash shall be distributed to holders of Allowed Claims under the Plan.

7.16.2 <u>Plan Administrator's Obligations Under Plan</u>. From and after the Effective Date, the Plan shall be administered and actions shall be taken in the name of Post Effective Date L&H through, in accordance with the terms hereof, the terms of the Plan Administration Agreement and the laws of the United States, the Plan Administrator, including, but not limited to, (a) administering the Plan and taking all steps and executing all instruments and documents necessary to effectuate the Plan and (b) administering the Distributions under the Plan, including (i) making Distributions in accordance with the terms of the Plan and (ii) establishing and maintaining the various Disputed Claims Reserves.

Securities Law Matters. It is an integral and essential element of this Plan 7.17 that (a) the Distribution to holders of Allowed Claims of the Litigation Trust Beneficial Interests pursuant to this Plan, (b) the distribution of the common stock of Reorganized Dictaphone distributed previously to L&H NV pursuant to the terms of the Dictaphone Plan and transferred to Post Effective Date L&H pursuant to the terms of this Plan, and (c) the liquidation, sale, Transfer or disposition of such common stock of Reorganized Dictaphone shall be exempt from registration under the Securities Act of 1933, as amended, pursuant to section 1145 of the Bankruptcy Code. Any such securities issued to an "affiliate" of the Debtor within the meaning of the Securities Act of 1933, or any Person that Post Effective Date L&H reasonably determines to be an "underwriter," and which does not agree to resell such securities only in "ordinary trading transactions," within the meaning of section 1145(b)(1) of the Bankruptey Code, shall be subject to such transfer restrictions and bear such legends as shall be appropriate to ensure compliance with the Securities Act of 1933. It also is an integral and essential element of the Plan that Rule 144 under the Securities Act of 1933 be available to any such "affiliate" that is not otherwise such an "underwriter" for purposes of permitting re-sales of such securities. In accordance with the Dictaphone Plan, solely for purposes of section 1145 of the Bankruptcy Code, Reorganized Dictaphone shall not be determined to be a "successor" of L&H NV.



MAR. 20. 2007 5:32PM

# SECTION VIII

#### DISTRIBUTIONS UNDER THE PLAN

8.1 <u>Timing of Distributions</u>. If any payment or act under the Plan is required to be made or performed on a date that is not a Business Day, then the making of such payment or the performance of such act may be completed on the next succeeding Business Day, but shall be deemed to have been completed as of the required date.

Delivery of Distributions. Subject to Bankruptcy Rule 9010, and except 8.2 as otherwise provided herein, Distributions to holders of Allowed Claims shall be made at the address of each of such holders as set forth in the Schedules filed with the Bankruptcy Court unless superseded by the address set forth on proofs of Claim filed by such holders (or at the last known address of such holders if no proof of Claim is filed, or if Post Effective Date L&H has been notified in writing of a change of address). If any Distribution to any holder of an Allowed Unsecured Claim is returned as undeliverable, no Distribution shall be made unless and until Post Effective Date L&H has determined the then-current address of such holder, at which time such Distribution to such holder shall be made to such holder without interest. Amounts in respect of any undeliverable Distributions made through Post Effective Date L&H shall be returned to and held by Post Effective Date L&H until such Distributions are claimed. Cash and other Distributions that are not claimed before the earlier of one year from the date of the Distribution or the date that the Chapter 11 Case is closed shall (a) be deemed unclaimed property under section 347(b) of the Bankruptcy Code, (b) shall revest in Post Effective Date L&H (and, shall be subject to redistribution, as appropriate, in accordance with the provisions of Section V of the Plan) and (c) shall be forever barred. Nothing contained in the Plan shall require Post Effective Date L&H to attempt to locate any holder of an Allowed Claim.

## 8.3 <u>Record Date</u>.

8.3.1 Equity Interests. At the close of business on the Record Date, the consolidated stockholders list or transfer ledger for the Equity Interests shall be closed, and there shall be no further changes in the record holders of the Equity Interests. Post Effective Date L&H, its agents and servicers shall have no obligation to recognize any transfer of such Equity Interests occurring after the Record Date. Post Effective Date L&H, its agents and its servicers shall be entitled instead to recognize and deal for all purposes hereunder with only (a) those record holders stated on the consolidated stockholders list or transfer ledger as of the close of business on the Record Date and (b) holders of Equity Interests domiciled outside of the United States who do not appear on the consolidated stockholders' list or transfer ledger and who held such Equity Interests as of the Record Date.

8.3.2 <u>Record Date for Holders of Claims</u>. Except as otherwise provided in a Final Order that is not subject to any stay, the transferees of C laims that are transferred pursuant to Bankruptcy Rule 3001 on or prior to the Record Date shall be treated as the holders of such Claims for all purposes, notwithstanding that any period provided by Bankruptcy Rule 3001 for objecting to such transfer has not expired by the Record Date.



8.4 <u>Time Bar to Cash Payments by Check</u>. Checks issued by Post Effective Date L&H on account of Allowed Claims shall be null and void if not negotiated prior to the earlier of one year from the date of such check or the closing of the Chapter 11 Case. Requests for re-issuance of any check shall be made in writing directly to Post Effective Date L&H by the holder of the Allowed Claim with respect to which such check originally was issued on or before the closing of the Chapter 11 Case. After such dates, all Claims in respect of void checks shall be forever barred, and the proceeds of such checks shall revest in and become the property of Post Effective Date L&H and subject to redistribution, as appropriate, in accordance with the provisions of Section V of the Plan.

8.S Disputed Claims Reserves. On the Effective Date or such later date that Distributions are required to be made on account of Allowed Claims, and after making all Distributions required to be made on any such date under the Plan, Post Effective Date L&H shall establish a separate Disputed Claims Reserve for each of the Classes receiving Distributions under the Plan, each of which Disputed Claims Reserves shall be administered by Post Effective Date L&H. Post Effective Date L&H shall reserve the Ratable Proportion of all Cash, Available Cash, or other Distributions allocated for each Disputed Claim, or such amount as may be agreed by the holder of such Claim and Post Effective Date L&H liable on such Claim, or as may otherwise be determined by order of the Bankruptcy Court. All Cash, Available Cash, or other Distributions, as applicable, allocable to the relevant Class hereunder shall be distributed by Post Effective Date L&H to the Disputed Claims Reserve on the Effective Date or such later date that Distributions are required to be made on account of Allowed Claims. The Disputed Claims Reserve shall be closed and extinguished by Post Effective Date L&H upon its determination that all Distributions and other dispositions of Cash, Available Cash, or other Distributions required to be made under the Plan have been made in accordance with the terms of the Plan. Upon closure of a Disputed Claims Reserve, all Cash and Available Cash (including any Cash investment Yield and any Cash dividends and other Distributions held in such Disputed Claims Reserve) shall be subject to redistribution, as appropriate, in accordance with the provisions of Section V of the Plan.

8.6 <u>Tax Requirements for Income Generated by Disputed Claims Reserves</u>. Post Effective Date L&H shall pay, or cause to be paid, out of the funds held in any of its Disputed Claims Reserves, any tax imposed by any federal, state or local taxing authority on the income generated by the funds or property held in such Disputed Claims Reserve. Post Effective Date L&H shall file, or cause to be filed, any tax or information return related to its Disputed Claims Reserves that is required by any federal, state or local taxing authority.

8.7 <u>Untimely Claims</u>. Except as otherwise expressly provided in this Plan, any Claim not deemed filed pursuant to section 1111(a) of the Bankruptcy Code or timely filed pursuant to the Bankruptcy Code, the Bankruptcy Rules, or any applicable order of the Bankruptcy Court, shall (a) not be treated as an Allowed Claim, and (b) be expunged from the Claims register in the Chapter 11 Case without need for any further notice, motion, or order.

8.8 <u>Estimation of Claims</u>. Post Effective Date L&H may, at any time, request that the Bankruptcy Court estimate any contingent or unliquidated administrative expense claim or Claim, including any Belgian Claim or Claim for taxes, to the extent permitted by section

33

502(c) of the Bankruptcy Code regardless of whether L&H NV, the Committee or Post Effective Date L&H have previously objected to such claim or whether the Bankruptcy Court has ruled on any such objection, and the Bankruptcy Court shall retain jurisdiction to estimate any claim at any time during litigation concerning any objection to any claim, including during the pendency of any appeal relating to any such objection. In the event that the Bankruptcy Court estimates any contingent or unliquidated claim, that estimated amount shall constitute either the Allowed amount of such claim or a maximum limitation on such claim, as determined by the Bankruptcy Court. If the estimated amount constitutes a maximum limitation on such claim. Post Effective Date L&H may elect to pursue supplemental proceedings to object to any ultimate allowance of such claim. All of the aforementioned claims objection, estimation and resolution procedures are cumulative and not necessarily exclusive of one another. Claims may be estimated and subsequently court.

8.9 <u>Distributions After Effective Date</u>. Distributions made after the Effective Date shall be decined to have been made on the Effective Date.

8.10 <u>Fractional Shares</u>. Notwithstanding any other provision of the Plan to the contrary, no fractional shares shall be issued pursuant to the Plan. Whenever any payment of a fraction of a share under the Plan would otherwise be required, the actual Distribution made shall reflect a rounding of such fraction to the nearest whole share (up or down), with half shares or more being rounded up and fractions less than half of a share being rounded down.

8.11 <u>Fractional Cents</u>. Notwithstanding any other provision of the Plan to the contrary, no payment of fractional cents shall be made pursuant to the Plan. Whenever any payment of a fraction of a cent under the Plan would otherwise be required, the actual Distribution made shall reflect a rounding of such fraction to the nearest whole penny (up or down), with half cents or more being rounded up and fractions less than half of a cent being rounded down.

8.12 <u>Minimum Distributions</u>. Notwithstanding anything herein to the contrary, if a Distribution to be made to a holder of an Allowed Claim on any date for Distributions (other than the final Distribution Date) would be \$50.00 or less in the aggregate, no such Distribution will be made to that holder unless a request therefor is made in writing to Post Effective Date L&H no later than twenty (20) days after the Effective Date.

8.13 <u>Interest on Claims</u>. Except as specifically provided for in the Plan or the Confirmation Order, interest shall not accrue on Claims, and no holder of a Claim shall be entitled to interest accruing on or after the Petition Date on any Claim. Interest shall not accrue or be paid on any Disputed Claim in respect of the period from the Petition Date to the date a final Distribution is made thereon if and after such Disputed Claim becomes an Allowed Claim. Except as expressly provided herein, no pre-petition Claim shall be Allowed to the extent that it is for post-petition interest or other similar charges.

8.14 <u>No Distribution in Excess of Allowed Amount of Claim</u>. Notwithstanding anything to the contrary herein, no holder of an Allowed Claim shall receive in respect of such Claim any Distribution in excess of the Allowed Amount of such Claim.

8.15 <u>Setoffs</u>. Except as otherwise provided in the Plan, Post Effective Date L&H may, but shall not be required to, setoff against any Claim and the Distributions to be made pursuant to the Plan in respect of such Claim, any Claims of any nature whatsoever that the Debtor may have against the holder of such Claim, but neither the failure to do so nor the allowance of any Claim under the Plan shall constitute a waiver or release by L&H NV or Post Effective Date L&H of any right of setoff any of them may have against the holder of such Claim.

8.16 <u>Payment of Taxes on Distributions Received Pursuant to Plan</u>. All Persons and Entities that receive Distributions under the Plan shall be responsible for reporting and paying, as applicable, taxes on account of such Distributions.

8.17 <u>Allocation of Distributions</u>. Post Effective Date L&H intends first to allocate any and all Distributions made with respect to the Claims as a repayment of principal, with the excess, if any, thereafter being allocated as a repayment of accrued but unpaid interest.

#### SECTION IX

#### DISPUTED CLAIMS UNDER THE PLAN

9.1 <u>Objection Deadline</u>. As soon as practicable, but in no event later than one-hundred eighty (180) days after the Effective Date, unless otherwise ordered by the Bankruptcy Court, objections to Claims and Equity Interests shall be filed with the Bankruptcy Court and served upon the holders of each such Claim or Equity Interest to which objections are made,

9.2 Prosecution of Objections after Effective Date. On and after the Effective Date, except as to applications for allowances of Professional Fees or as otherwise ordered by the Bankruptcy Court, the filing, litigation, settlement, or withdrawal of all objections to Claims and Equity Interests, including pending objections, shall be the responsibility of Post Effective Date L&H. Any Claim, other than a Claim for Professional Fees, which is not an Allowed Claim shall be determined, resolved, or litigated by Post Effective Date L&H by and through the Plan Administrator. Prior to the Effective Date, the filing, litigation, settlement, or withdrawal of all objections shall be the responsibility of the Debtor.

9.3 <u>No Distributions Pending Allowance</u>. Notwithstanding any other provision hereof, if any portion of a Claim is a Disputed Claim, no payment or Distribution provided hereunder shall be made on account of the portion of such Claim that is a Disputed Claim unless and until such D isputed Claim b ecomes an Allowed Claim, but the payment or Distribution provided hereunder shall be made on account of the portion of such Claim that is an Allowed Claim.

9.4 <u>Withholding of Allocated Distributions</u>. Post Effective Date L &H shall withhold from the property to be distributed under the Plan for the benefit of holders of Disputed Claims Distributions in an amount sufficient to be distributed on account of such Disputed Claims, which Distributions shall be deposited in the applicable Disputed Claims Reserve.

9.5 <u>Distribution When a Disputed Claim Becomes an Allowed Claim</u>. Distributions to each holder of a Disputed Claim, to the extent that such Claim ultimately becomes an Allowed Claim (and to the extent that the holder of such Claim has not received prior Distributions on account of such Claim), shall be made in accordance with the provisions of the Plan governing the Class of Claims in which such Claim is classified.

# SECTION X

# EXECUTORY CONTRACTS AND UNEXPIRED LEASES UNDER THE PLAN

10.1 <u>All Executory Contracts and Unexpired Leases Rejected if Not Listed on</u> <u>Assumption Schedule</u>. Except as otherwise provided herein or pursuant to a Final Order approving the assumption and assignment of any executory contract or unexpired lease, effective as of the Confirmation Date, all executory contracts and unexpired leases of the Estate not specifically listed on the Assumption And Assignment Schedule shall be deemed to be automatically rejected as of the Confirmation Date. The Confirmation Order shall constitute an order of the Bankruptcy Court approving such rejections pursuant to section 365(a) of the Bankruptcy Code as of the Confirmation Date.

Assumed Executory Contracts and Unexpired Leases. 10.2All executory contracts and unexpired leases of the Estate specifically listed on the Assumption And Assignment Schedule shall be deemed automatically assumed and assigned to the party set forth in such list upon the Confirmation Date. The Confirmation Order shall constitute an order of the Bankruptcy Court approving such assumptions and assignments pursuant to section 365(a) of the Bankruptcy Code. The Committee may at any time on or before the Confirmation Date (or, with respect to any executory contracts and unexpired leases for which there is a dispute regarding the nature or the amount of any Cure, at any time on or before the date of the entry of a Final Order resolving such dispute) amend the Assumption And Assignment Schedule to delete therefrom or add thereto any executory contract or unexpired lease, in which event such executory contract or unexpired lease shall be deemed to be rejected or assumed, respectively, as of the Confirmation Date. The Committee shall provide notice of any amendments to the Assumption And Assignment Schedule to the parties to the executory contracts or unexpired leases affected thereby, counsel to the Debtor, and to parties who formally have requested notice pursuant to Bankruptcy Rule 2002. The fact that any contract or lease is listed on the Assumption And Assignment Schedule shall not constitute or be construed to constitute an admission that such contract or lease is an executory contract or unexpired lease within the meaning of section 365 of the Bankroptcy Code or that the Debtor or any successor in interest to the Debtor has any liability thereunder. Each assumed and assigned executory contract and unexpired lease of the Debtor that relates to the use or occupancy of real property shall include (a) all modifications, amendments, supplements, restatements, or other agreements made directly or indirectly by any agreement, instrument, or other document that in any manner affacts such executory contract or unexpired lease and (b) all executory contracts or unexpired leases appurtenant to the premises, including all easements, licenses, permits, rights, privileges, immunities, options, rights of first refusal, powers, uses, usufracts, reciprocal easement agreements, vaults, tunnel or bridge agreements, or franchises, and any other interests in real estate or rights in rem related to such promises, unless any of the foregoing agreements has been rejected pursuant to a Final Order.

10.3 <u>Payments Related to Assumption of Executory Contracts and Unexpired</u> <u>Leases</u>. A nymonetary amounts by which each executory contract and unexpired lease to be assumed and assigned under the Plan may be in default shall be satisfied by Cure, under section 365(b)(1) of the Bankruptcy Code. In the event of a dispute regarding (a) the nature or the amount of any Cure, (b) the ability of the assignee to provide "adequate assurance of future performance" (within the meaning of section 365 of the Bankruptcy Code) under the contract or lease to be assumed, or (c) any other matter pertaining to assumption, Cure shall occur following the entry of a Final Order resolving the dispute.

10.4 <u>Bar Date for Rejection Damages</u>. If the rejection by the Debtor, pursuant to the Plan, of an executory contract or unexpired lease results in a Claim or Administrative Expense Claim, then such Claim or Administrative Expense Claim shall be discharged and barred forever and shall not be enforceable against L&H NV, Post Effective Date L&H or the Chapter 11 Assets, unless a proof of Claim or proof of Administrative Expense Claim is filed with the clerk of the Bankruptcy Court and served upon Post Effective Date L&H within thirty (30) days after the earlier to occur of (a) the Confirmation Date and (b) the entry of an order by the Bankruptcy Court authorizing rejection of the subject executory contract or unexpired lease.

10.5 <u>Retiree Benefits</u>. L&H NV has no retiree benefit plans, fund, or programs, as defined in section 1114 of the Bankruptcy Code, for the purpose of providing or reimbursing payments for retired employees and their spouses and dependents for medical, surgical, or hospital care benefits, or benefits in the event of sickness, accident, disability, or death under any plan, fund or program (through the purchase of insurance or otherwise). Accordingly, no payments relating to such retiree benefit plans, fund, or programs shall be made.

# SECTION XI

## CONDITIONS PRECEDENT TO THE CONFIRMATION DATE AND THE EFFECTIVE DATE

11.1 <u>Conditions Precedent to the Confirmation of Plan</u>. The following are conditions precedent to confirmation of the Plan that must be satisfied, unless waived in accordance with Section 11.3 of the Plan:

(a) Entry of the Confirmation Order. The Confirmation Order shall be in form and substance acceptable to the Committee and shall, among other things:

(i) decree that the assets of Post Effective Date L&H shall be free and clear of all Claims, Liens and encombrances;

(ii) decree that all transfers of the Chapter 11 Assets contemplated under the Plan shall be free and clear of all Claims, Liens and all encumbrances against such assets and equity;

(iii) authorize the implementation of the Plan in accordance with its terms;

(iv) provide that any transfers effected or mortgages entered into or to be effected or entered into under the Plan shall be and are exempt from any state, city or other municipality transfer taxes, mortgage recording taxes and any other stamp or similar tax under section 1146(c) of the Bankruptcy Code;

(v) approve the other settlements, transactions and agreements to be effected pursuant to the Plan in all respects;

(vi) authorize the transfer of \$14,850,000 in Cash to the Curators for administration in the Belgian Case and for the benefit of Belgian Priority Claims in accordance with the terms of the Plan;

(vii) provide that all executory contracts or unexpired leases assumed by L&H NV and assigned during the Chapter 11 Case or order the Plan shall remain in full force and effect notwithstanding any provision in such contract or lease (including those provisions described in sections 365(b)(2) and (f) of the Bankruptcy Code) that prohibits such

assignment or transfer or that enables or requires termination of such contract or lease;

(viii) provide that all executory contracts or unexpired leases assumed by L&H NV during the Chapter 11 Case or under the Plan shall remain in full force and effect for the benefit of Post Effective Date L&H;

(ix) provide that the transfers of the Chapter 11 Assets to Post Effective Date L&H (A) are or shall be legal, valid, and effective transfers of property, (B) vest or shall vest Post Effective Date L&H with good title to such property free and clear of all Liens, Claims, encumbrances, and other interests, except as expressly provided in the Plan or Confirmation Order, (C) do not and shall not constitute avoidable transfers under the Bankruptcy Code or under applicable bankruptcy or non-bankruptcy law, and (D) except as expressly provided in the Plan, do not and shall not subject Post Effective Date L&H to any Jiability by reason of such transfer under the Bankruptcy Code or under applicable non-bankruptcy law, including, without limitation, any laws affecting successor or transferce liability;

(x) determine that any objection, not previously withdrawn or settled, to the adequacy of the information contained in the Disclosure Statement is overruled, and that the information contained in the Disclosure Statement was adequate for the purpose of soliciting votes for acceptance of the Plan;

(xi) find that the Plan complies with all applicable provisions of the Bankruptcy Code, including, without limitation, that the Plan was proposed in good faith and that the Confirmation Order was not procured by fraud;

(xii) except as expressly provided in the Plan, provide that all Equity Interests shall be extinguished and cancelled effective upon the Effective Date with respect to the Chapter 11 Assets;

(xiii) determine that the treatment of Scourities Laws Claims as provided in the Plan is approved and authorized in all respects, effective as of the Effective Date, and that such treatment is valid, appropriate and in the best interests of the Estates;

(xiv) decree that, in accordance with section 508(a) of the Bankruptcy Code, any creditor that receives any consideration on behalf of a claim in the Belgian Case shall be prohibited from receiving a Distribution on such claim until all other creditors with claims of equal priority receive a Distribution equal in value to the consideration received by such creditor in the Belgian Case; and

(xv) determine that the allocation of L&H NV's assets between the Chapter 11 Case and the Belgian Case is fair, reasonable, approved, appropriate and in the best interests of the Estate.

(b) Finality of Disclosure Statement Approval Order. The Disclosure Statement Approval Order shall have been entered and be a Final Order.

11.2 <u>Conditions Procedent to the Effective Date of the Plan</u>. The Plan shall not become effective and the Effective Date shall not occur unless and until the following conditions shall have been satisfied or waived in accordance with Section 11.3 of the Plan:

11.2.1 <u>Confirmation Order</u>. The Confirmation Date shall have occurred and the Confirmation Order shall have been signed by the judge presiding over the Chapter 11 Case, and shall have become a Final Order.

11.2.2 <u>Conditions to the Confirmation Date Remain Satisfied</u>. All conditions precedent to the Confirmation Date shall have been satisfied and shall continue to be satisfied.

11.2.3 Execution of Documents. All actions, documents and agreements necessary to implement the provisions of the Plan to be effectuated on or prior to the Effective Date shall be reasonably satisfactory to the Committee and such actions, documents and agreements shall have been effected or executed and delivered.

11.2.4 <u>Release of Collateral</u>. Unless a particular Secured Claim is Reinstated: (i) each holder of: (A) a Secured Claim; and/or (B) a Claim that is purportedly secured shall on or immediately before the Effective Date: (x) turn over and release to Post Effective Date L&H any and all properly of Post Effective Date L&H that secures or purportedly secures such Claim; and (y) execute such documents and instruments as Post Effective Date requires to evidence such claimant's release of such property; and (ii) on the Effective Date, all claims, right, title and interest in such property shall vest in Post Effective Date L&H free and clear of all Claims and Equity Interests, including, without limitation, Liens, charges, pledges, interests, encumbrances and/or security interests of any kind.

11.3 <u>Waiver of Conditions Precedent</u>. Each of the conditions precedent in Sections 11.1 and 11.2 hereof may be waived or modified, in whole or in part by the Committee. Any such waiver or modification of a condition precedent in Sections 11.1 and 11.2 hereof may be effectuated at any time, without notice, without leave or order of the Bankruptcy Court and without any other formal action,

11.3.1 Effect of Failure or Absence of Waiver of Conditions Precedent to the Effective Date of the Plan. In the event that one or more of the conditions specified in Section 11.2 of the Plan have not occurred (or been waived) on or before 120 days after the Confirmation Date, upon notification submitted by the Committee to the Bankruptcy Court and the Debtor, (a) the Confirmation Order shall be vacated, (b) no distributions under the Plan shall be made, (c) the Debtor and all holders of Claims and Equity Interests shall be restored to the status quo ante as of the day immediately preceding the Confirmation Date as though the

Confirmation Date never occurred and (d) the Debtor's obligations with respect to the Claims and Equity Interests shall remain unchanged and nothing contained herein shall constitute or be deemed a waiver or release of any Claims or Equity Interests by or against the Debtor or any other Person or Entity or to prejudice in any manner the rights of the Debtor or any Person or Entity in any further proceedings involving the Debtor.

#### SECTION XII

#### **EFFECT OF CONFIRMATION**

12.1 Post Effective Date L&H's Authority. Until the Effective Date, the Bankruptcy Court shall retain custody of and jurisdiction over the Debtor, including the Chapter 11 Assets. On the Effective Date, after transferring the Chapter 11 Assets to Post Effective Date L&H and the Assigned Causes if Action to the Litigation Trust, L&H NV, its property and interests in property and operations shall be released from the custody and jurisdiction of the Bankruptcy Court, except as otherwise provided herein.

12.2 <u>Vesting and Liens</u>. On the Effective Date, all of the Chapter 11 Assets shall be vested in Post Effective Date L&H free and clear of all Liens (except as expressly provided herein).

12.3 Injunction, Except as otherwise provided in the Plan or Confirmation Order (including any right to receive Distributions under the Plan) or a separate order of the Bankruptcy Court, as of the Effective Date, all Entities that have held, currently hold or may hold a Claim or other debt or liability that would be discharged or an Equity Interest or other right of an equity security holder that is terminated and canceled pursuant to the terms of the Plan, are permanently enjoined from taking any of the following actions on account of any such discharged Claims, debts or liabilities or terminated and canceled Equity Interests or rights: (1) commencing or continuing in any manner any action or other proceeding against Post Effective Date L&H, the Chapter 11 Assets or properties and interests in properties of each of the foregoing; (2) enforcing, attaching, collecting or recovering in any manner any judgment, award, decree or order against Post Effective Date L&H, the Estate or properties and interests in properties of each of the foregoing; (3) creating, perfecting or enforcing any lien or encumbrance against Post Effective Date L&H, the Chapter 11 Assets or properties and interests in properties of each of the foregoing; (4) asserting a setoff, right of subrogation or recoupment of any kind against any obligation due to Post Effective Date L&H, the Chapter 11 Assets or properties and interests in properties of each of the foregoing; and (5) commencing or continuing any action, in any manner, in any place that does not comply with or is inconsistent with the provisions of the Plan and the Confirmation Order. In addition, all holders of Belgian Priority Claims shall be prohibited from asserting such claims in the Chapter 11 Case or seeking consideration pursuant to the Plan on account of a Belgian Priority Claim. Such injunction shall extend to all successors of the Debtor and its creditors. Nothing in the Plan or the Confirmation Order shall prohibit, impede or prevent any party from pursuing a claim or taking any other action in the Belgian Case, or from pursuing a claim or taking any action against any assets other than the Chapter 11 Assets.

Nothing in the Plan or the Confirmation Order shall be deemed to discharge, enjoin, restrict or otherwise impair any rights that may exist in favor of a person or entity to assert any defensive rights of scioff or recoupment with respect to any cause of action that may be asserted against such person or entity by the Debtor or successor in interest to the Debtor, including the Litigation Trust described herein.

12.4 <u>Term of Injunctions or Stays</u>. Unless otherwise provided herein or in the Confirmation Order, all injunctions or stays provided for in the Chapter 11 Case pursuant to section 105 or 362 of the Bankruptcy Code, or otherwise, and in existence on the Confirmation Date, shall remain in full force and effect until the Effective Date.

## SECTION XIII

## EXCULPATION

13.1 <u>Avoidance and Recovery Actions</u>. In accordance with Section 7.4.2 of the Plan, and subject to Section 7.14 of the Plan, all of the Assigned Causes of Action that belong to or could have been raised by or on behalf of the Debtor or Debtor In Possession or its respective Estate, have been transferred and assigned to the Litigation Trust. Post Effective Date L&H and the Litigation Trust, as the case may be, as the successors to the Debtor, shall retain and may prosecute any of the foregoing as a defense or counterclaim to any Claim, Counterclaim or action, including, but not limited to, any rights under section 502(d) of the Bankruptcy Code. All other Causes of Action are being retained by Post Effective Date L&H in accordance with Section 7.14 of the Plan.

13.2 Exculpation. Except to the extent of any willful misconduct or gross negligence, none of L&H NV, Post Effective Date L&H, or any of the L&H NV Parties (but solely in their capacities as L&H NV Parties) shall have or incur any liability to any holder of a Claim or Equity Interest for any act or omission in connection with, related to, or arising out of, (a) the Chapter 11 Case, (b) the pursuit of confirmation of the Plan, (c) the consummation of the Plan or the administration of the Plan or the property to be distributed under the Plan, (d) the Plan, (e) the negotiation, formulation and preparation of the Plan and any of the terms, settlements and compromises reflected in the Plan, and, in all respects, Post Effective Date L&H and each of the L&H NV Parties shall be entitled to rely upon the advice of counsel with respect to their duties and responsibilities under the Plan; provided, however, that the exculpation contained in this Section 13.2 of the Plan shall not apply, (x) with respect to the L&H NV Parties, to acts or omissions that occurred prior to the Petition Date, and (y) solely with respect to Post Effective Date L&H, the Litigation Trust, the Plan Administrator, the Litigation Trustee or the Curators, to acts or omissions that occur after the Effective Date.

13.3 <u>Release Of Certain Intercompany Claims</u>. As of the Effective Date, L&H NV, Post Effective Date L&H, and each of their respective successors and successors-in-interest, hereby release any and all Claims, other claims or Causes of Action that they have, may have, or claim to have, which are property of, assertable on behalf of, or derivative of the Estate against L&H Holdings or its Affiliates (other than L&H NV) in accordance with that certain Mutual Release Agreement among L&H Holdings and L&H NV dated on or about June 26, 2002, which

was approved by the Bankraptey Court on August 13, 2002 and which became effective on the effective date of the L&H Holdings Plan; provided, however, such mutual releases among L&H NV and L&H Holdings shall not include (a) a release of certain claims relating to intercompany allocations of shared costs, including management costs, corporate markering costs, professional fees and expenses, DIP fees, interest and related expenses, and insurance costs (which allocations have or shall be approved by the Bankruptcy Court) and (b) the U.S. Merger Claims and the Other U.S. Claims assigned to L&H NV pursuant to the Stipulation And Order Between Lornout & Hauspie Speech Products N.V. And L&H Holdings USA, Inc. And Baker Parties Compromising And Settling Claims (as such Claims are defined therein), dated January 31, 2002; and provided, further, that such releases shall include, but not be limited to, (x) other than the Claims provided for in subsection (b) above, any and all claims relating to that certain Agreement And Plan Of Merger dated as of March 27, 2000, by and among L&H NV, L&H Holdings, Dragon Systems, Inc. and the principal stockholders of Dragon Systems, Inc., any claims of breach of warranty or representations relating thereto, and (y) any remaining prepetition and post-petition intercompany claims among L&H NV.

#### SECTION XIV

#### **RETENTION OF JURISDICTION**

14.1 <u>Retention of Jurisdiction</u>. The Bankruptcy Court may retain jurisdiction; and if the Bankruptcy Court exercises its retained jurisdiction, shall have exclusive jurisdiction, of all matters arising out of, and relating to, the Chapter 11 Case and the Plan pursuant to, and for the purposes of, sections 105(a) and 1142 of the Bankruptcy Code and for, among other things, the following purposes:

> (i) To hear and determine pending motions for the assumption or rejection of executory contracts or unexpired leases, if any are pending, and the allowance of Claims resulting therefrom;

 (ii) To determine any and all adversary proceedings, applications and contested matters;

(iii) To consider and rule on the compromise and settlement of any Claim against or Cause of Action on behalf of Post Effective Date L&H or its Estate;

(iv) To ensure that Distributions to holders of Allowed Claims are accomplished as provided herein;

(v) To hear and determine any timely objections to Administrative Expense Claims, Belgian Claims or to proofs of Claim filed, both before and after the Confirmation Date, including any objections to the classification of any Claim, and to Allow or Disallow any Disputed Claim, in whole or in part;

(vi) To hear and determine any and all applications for the allowance of Professional Fees;

(vii) To enter and implement such orders as may be appropriate in the event the Confirmation Order is for any reason stayed, revoked, modified, or vacated;

(viii) To issue such orders in aid of execution of the Plan, in accordance with section 1142 of the Bankruptcy Code;

(ix) To consider any modifications of the Plan, to cure any defect or omission, or reconcile any inconsistency in the Plan, including any exhibit thereto, or in any order of the Bankruptoy Court, including the Confirmation Order, as may be necessary to carry out the purposes and intent of the Plan and to implement and effectuate the Plan;

(x) To hear and determine all applications for awards of compensation for services rendered and reimbursement of expenses relating to implementation and effectuation of the Plan;

(xi) To hear and determine disputes arising in connection with the interpretation, implementation or enforcement of the Plan, including, but not limited to, disputes relating to the Plan Transactions;

(xii) To hear and determine matters concerning state, local and federal taxes in accordance with sections 346, 505 and 1146 of the Bankruptey Code;

(xiii) To compel the conveyance of property and other performance contemplated under the Plan and documents executed in connection herewith;

(xiv) To enforce remedies upon any default under the Plan;

(xv) To enforce, interpret and determine any disputes arising in connection with any orders, stipulations, judgments and rulings entered in connection with the Chapter 11 Case (whether or not the Chapter 11 Case has been closed);

(xvi) To resolve any cases, controversies, suits or disputes that may arise in connection with the consummation, interpretation or enforcement of the Plan, or any Person's or Entity's obligations incurred in connection herewith;

(xvii) To determine any other matters that may arise in connection with or relate to the Plan, the Disclosure Statement, the Confirmation Order or any contract, instrument, release, indenture or other

agreement or document created in connection with the Plan (including, but not limited to, the Plan Supplement) or the Disclosure Statement;

(xviii) To issue injunctions, enter and implement other orders or take such other actions as may be necessary or appropriate to restrain interference by any Person or Entity with the occurrence of the Effective Date or enforcement of the Plan;

(xix) To issue such orders as may be necessary or appropriate in aid of confirmation and/or to facilitate consummation of the Plan;

(xx) To determine such other matters as may be provided for in the Confirmation Order or other orders of the Bankruptcy Court as may be authorized under the provisions of the Bankruptcy Code or any other applicable law;

(xxi) To hear and determine (a) all motions, applications, adversary proceedings, and contested and higgated matters pending on the Effective Date, and (b) all claims by or against the Chapter 11 Assets arising under the Bankruptcy Code or non-bankruptcy law, if made applicable by the Bankruptcy Code, whether such claims are commenced before or after the Effective Date;

(xxii) To hear any other matter not inconsistent with the Bankruptcy Code; and

(xxiii) To enter a final decree closing the Chapter 11 Case.

14.2 <u>Modification of Plan.</u>

(a) Subject to Sections 6.5 and 15.8 of the Plan, the Committee may alter, amend, or modify the Plan or any Exhibits thereto under section 1127(a) of the Bankrupicy Code at any time prior to the Confirmation Date. The Committee shall provide partics-in-interest with notice of such amendments or modifications as may be required by the Bankrupicy Rules or order of the Bankrupicy Court. A holder of a Claim that has accepted the Plan shall be deemed to have accepted the Plan, as altered, amended, modified or clarified, if the proposed alteration, amendment, modification or clarification does not materially and adversely change the treatment of the Claim of such holder.

(b) After the Confirmation Date and prior to substantial consummation (as defined in section 1101(2) of the Bankruptcy Code) of the Plan, Post Effective Date L&H may, under section 1127(b) of the Bankruptcy Code, institute proceedings in the Bankruptcy Court to remedy any defect or omission or to reconcile any inconsistencies in the Plan, the Disclosure Statement approved with respect to the Plan, or the Confirmation Order, and such matters as may be necessary to carry out the purpose and effect of the Plan so long as such proceedings do not adversely affect the treatment of holders of Claims under the Plan; provided, however, that, to the extent required, prior notice of such proceedings shall be served in accordance with the

Bankruptcy Rules or an order of the Bankruptcy Court. A holder of a Claim that has accepted the Plan shall be deemed to have accepted the Plan, as altered, amended, modified or clarified, if the proposed alteration, amendment, modification or clarification does not materially and adversely change the treatment of the Claim of such holder.

# SECTION XV

## MISCELLANEOUS PROVISIONS

15.1 <u>Payment of Statutory Fees</u>. All fees payable pursuant to section 1930, title 28, United States Code, shall be paid on the Effective Date, and thereafter as they come due pending closing or dismissal of the case.

15.2 Role of Indenture Trustees: Certain Fees And Expenses of Indenture Trustees. Notwithstanding any other provision in this Plan, as soon as practicable after the Record Date, the Old Convertible Subordinated Notes Indenture Trustee and the PIERS Indenture Trustee shall provide to Post Effective Date L&H a list setting forth the identities of holders of any Class 4 PIERS/Old Convertible Subordinated Notes Claim and such other information that Post Effective Date L&H may require to effect such Distributions to holders of such Claims. Subject to the Effective Date occurring, the reasonable fees and expenses of the Old Convertible Subordinated Notes Indenture Trustee and the PIERS Indenture Trustee for services rendered shall be Allowed as Administrative Expense Claims against L&H NV, pursuant to section 503(b) of the Bankruptcy Code, and shall be paid by L&H NV or Post-Effective Date L&H without the need for the Old Convertible Subordinated Notes Trustee and the PIERS Indenture Trustee to file an application for allowance with the Bankruptcy Court, subject to such fees and expenses being reasonable. The Bankruptcy Court shall retain jurisdiction over any dispute regarding the reasonableness of such focs and expenses, and any and all rights, claims, and defenses of L&H NV and Post-Effective Date L&H with respect to the reasonableness of such fees and expenses, including, but not limited to the right to object to such fees and expenses, shall be expressly reserved.

15.3 Governing Law. EXCEPT TO THE EXTENT THAT THE BANKRUPTCY CODE OR OTHER FEDERAL LAW IS APPLICABLE, OR TO THE EXTENT A SCHEDULE OR EXHIBIT HERETO OR INSTRUMENT, AGREEMENT OR OTHER DOCUMENT EXECUTED UNDER THE PLAN PROVIDES OTHERWISE, THIS PLAN, THE RIGHTS, DUTIES AND OBLIGATIONS ARISING UNDER THIS PLAN, AND ANY CLAIM OR CONTROVERSY DIRECTLY OR INDIRECTLY BASED UPON OR ARISING OUT OF THIS PLAN OR THE TRANSACTIONS CONTEMPLATED BY THIS PLAN (WHETHER BASED ON CONTRACT, TORT, OR ANY OTHER THEORY), INCLUDING ALL MATTERS OF CONSTRUCTION, VALIDITY AND PERFORMANCE, SHALL IN ALL RESPECTS BE GOVERNED BY AND INTERPRETED, CONSTRUED AND DETERMINED IN ACCORDANCE WITH, THE INTERNAL LAWS OF THE STATE OF NEW YORK (WITHOUT REGARD TO ANY CONFLICTS OF LAW PROVISION THAT WOULD REQUIRE THE APPLICATION OF THE LAW OF ANY OTHER JURISDICTION).



46

15.4 <u>Notices</u>. All notices, requests and demands to be effective shall be in writing (including by facsimile transmission) and, unless otherwise expressly provided herein, shall be deemed to have been duly given or made when actually delivered or, in the case of notice by facsimile transmission, when received and telephonically confirmed, addressed as follows:

To Counsel for the Committee:

Akin Gump Strauss Hauer & Feld LLP 590 Madison Avenue, 20th Floor New York, New York 10022 Attn: Ira S. Dizengoff, Esq. James R. Savin, Esq. Telephone: (212) 872-1000 Facsimile: (212) 872-1002

-and-

Monzack and Monaco 400 Commerce Center Twelfth and Orange Streets Wilmington, Delaware 19899 Attn: Francis A. Monaco, Jr., Esq. Joseph Bodnar, Esq. Telephone: (302) 656-8162 Faesimile: (302) 656-2769

To Post Effective Date L&H/Plan Administrator/Litigation Trustee:

Scott L. Baena, Esq. c/o Bilzin Sumberg Baena Price & Axelrod LLP Wachovia Financial Center 200 South Biscayne Blvd, Suite 2500 Miami, FL 33131 Telephone: (305) 374-7580 Facsimile: (305) 374-7593



To L&H NV:

Lemout & Hausple Speech Products N.V. C/o Jean-Mare Vanstaen, Nieuwstraat 23 Wervik, Belgium Telephone: 011-32-56-31-14-06 Facsimile: 011-32-56-31-40-62

with a copy to Counsel for L&H NV:

Milbank, Tweed, Hadley & McCloy LLP 1 Chase Manhattan Plaza New York, NY 10005-1413 Attn: Lnc A. Despins, Esq. Maithew S. Barr, Esq. James C. Tecce, Esq. Telephone: (212) 530-5398 Facsimile: (212) 530-5219

-and-

Morris, Nichols, Arsht & Tunnell 1201 North Market Street P.O. Box 1347 Wilmington, Delaware 19899-1347 Attn: Robert J. Dehney, Esq. Gregory W. Werkheiser, Esq. Donna L. Harris, Esq. Telephone: (302) 658-9200 Facsimile: (302) 658-3989

To the Lenders:

KBC Bank N.V. 125 West 55th Street New York, NY 10019 Attn: Michael Curran Telephone: (212) 541-0600 Facsimile: (212) 541-0784



with a copy to Counsel for the Lenders:

Chadbourne & Parke LLP 30 Rockefeiler Plaza New York, NY 10112 Attn: Howard Seife, Esq. Theodore Zink, Esq. Telephone: (212) 408-5100 Facsimile: (212) 541-5369

To the United States Trustee:

The Office of the United States Trustee J. Caleb Boggs Federal Building Suite 2313, Loxbox 35 Wilmington, DE 19899 Attn: Mark Kenney, Esq. Telephone: (302) 573-6491 Facsimile: (302) 573-6497

15.5 <u>Further Documents and Action</u>. Post Effective Date L&H shall execute, and is authorized to file with the Bankruptcy Court, such agreements and other documents, take or cause to be taken such action, and deliver such documents or information as may be necessary or appropriate to effect and further evidence the terms and conditions of the Plan and to consummate the transactions and transfers contemplated by the Plan. Post Effective Date L&H, and all other parties, shall execute any and all documents and instruments that must be executed under or in connection with the Plan in order to implement the terms of the Plan or to effectuate the Distributions under the Plan, provided that such documents and instruments are reasonably acceptable to such party or parties.

15.6 <u>Plan Supplement</u>. Except as otherwise provided in the Plan, forms of the following documents shall be contained in the Plan Supplement and filed with the Clerk of the Bankruptcy Court at least seven (7) days prior to the Voting Deadline: the Assumption And Assignment Schedule, the Plan Administration Agreement and the Litigation Trust Agreement, the Mutual Release Agreement among L&H NV and L&H Holdings dated on or about June 26, 2092, and if available by such date, the lists of the initial members of the Litigation Monitoring Committee. Upon its filing with the Bankruptcy Court, the Plan Supplement may be inspected in the office of the Clerk of the Bankruptcy Court during normal Bankruptcy Court hours. Holders of Claims may obtain a copy of the Plan Supplement upon written request to the Committee in accordance with Section 15.4 of the Plan.

15.7 <u>Plan Controls: Conflicts</u>. In the event of any conflict or inconsistency between the terms of (a) the Plan (including all exhibits to the Plan), and (b) the Disclosure Statement, the terms of the Plan shall control.



PATENT REEL: 019047 FRAME: 0214

49

15.8 <u>Reservation of Rights</u>. If the Plan is not confirmed by a Final Order, or if the Plan is confirmed and does not become effective, the rights of all parties in interest in the Chapter 11 Case are and shall be reserved in full. Any concessions or settlements reflected herein, if any, are made for purposes of the Plan only, and if the Plan does not become effective, no party in interest in the Chapter 11 Case shall be bound or deemed prejudiced by any such concession or settlement.

15.9 <u>Tax Reporting and Compliance</u>. In connection with the Plan and all instruments issued in connection therewith and Distributions thereon, Post Effective Date L&H shall comply with all withholding and reporting requirements imposed by any federal, state, local or foreign taxing authority and all Distributions hereunder shall be subject to any such withholding and reporting requirements. No holder of an Allowed Claim or Equity Interest against the Chapter 11 Assets shall effectuate any withholding with respect to the cancellation or satisfaction of such Allowed Claim under the Plan. Post Effective Date L&H is hereby authorized to request an expedited determination of taxes under section 505(b) of the Bankruptcy Code for all taxable periods of L&H NV ending after the Petition Date through, and including, the Effective Date of the Plan.

15.10 <u>Binding Effect</u>. The rights, benefits and obligations of any Entity named or referred to in the Plan, or whose actions may be required to effectuate the terms of the Plan shall be binding on, and shall inure to the benefit of, any heir, executor, administrator, successor er assign of such Entity (including, but not limited to, any trustee appointed for L&H NV under Chapter 7 or 11 of the Bankruptey Code). The Confirmation Order shall provide that the terms and provisions of the Plan and the Confirmation Order shall survive and remain effective after entry of any order which may be entered converting the Chapter 11 Case to a case under Chapter 7 of the Bankruptey Code, and the terms and provisions of the Plan shall continue to be effective in this or any superseding case under the Bankruptey Code.

[END OF TEXT]

IN WITNESS WHEREOF, the undersigned has duly executed the Plan as of April 10, 2003.

# OFFICIAL COMMITTEE OF UNSECURED CREDITORS OF LERNOUT & HAUSPIE SPEECH PRODUCTS N.V.

By: <u>/s/</u>	Michael Corran
Name:	Michael Curran on behalf of KBC Bank N.V.
Capacity:	Chairman of the Official Committee of
	Unsecured Creditors of Lemout & Hauspie
	Speech Products N.V.

\$I

# Exhibit B

Notice to Filing and Recording Officers of Entry and Terms of Confirmation Order

#### IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

■ = = = = = = = = = = = = = = = = = = =	x
În re:	÷
LERNOUT & HAUSPIE	;
SPEECH PRODUCTS N.V.,	;
	:
Debtor.	;
	х

Chapter 11

Case No. 00-4398 (JHW)

# NOTICE TO FILING AND RECORDING OFFICERS OF ENTRY AND TERMS OF CONFIRMATION ORDER

#### TO ALL FILING AND RECORDING OFFICERS:

PLEASE TAKE NOTICE that on May 29, 2003, the United States Bankruptcy Court for the District of Delaware entered the Findings of Fact and Conclusions of Law Relating To, And Order Under 11 U.S.C. § 1129 Confirming, The Official Committee of Unsecured Creditors' Plan of Liquidation for Lernout & Hauspie Speech Products N.V. under Chapter 11 of the Bankruptcy Code, dated March 11, 2003 (the "<u>Confirmation Order</u>"). (Capitalized terms not defined in this notice have the meanings ascribed to them in the Confirmation Order or Plan). A copy of the Confirmation Order and the Plan are annexed hereto.

PLEASE TAKE FURTHER NOTICE that paragraph 38 of the Confirmation Order provides as follows:

(a) Pursuant to Section 1146(c) of the Bankruptcy Code, the issuance, transfer, or exchange of any security, or the making, delivery, filing, or recording of any instrument of transfer, under the Plan shall not be taxed under any law imposing a stamp tax, transfer tax, or similar tax.

(b) Without limiting the generality of subparagraph (a) of this paragraph, the making, delivery, filing, or recording at any time of any deed, bill of sale, mortgage, leasehold mortgage, deed of trust, leasehold deed of trust, memorandum of lease, notice of lease, assignment, leasehold assignment, security agreement, financing statement, or other instrument of absolute or collateral transfer required, or deemed necessary or desirable, by L&H NV, Post Effective Date L&H or the Plan Administrator, and other agreements or instruments related thereto shall not be so taxed.

(c) All filing or recording officers, wherever located and by whomever appointed, are hereby directed to accept for filing or recording, and to file or record immediately upon presentation thereof, all such deeds, bills of sale, mortgages, leasehold mortgages, deeds of trust, leasehold deeds of trust, memoranda of lease, notices of lease, assignments, leasehold assignments, security agreements, financing statements, and other instruments of absolute or collateral transfer without payment of any stamp tax, transfer

tax, or similar tax imposed by federal, state, or local law. This Notice, (i) shall have the effect of an order of this Court, (ii) shall constitute sufficient notice of the entry of this Order to such filing and recording officers, and (iii) shall be a recordable instrument notwithstanding any contrary provision of nonbankruptcy law. This Court specifically retains jurisdiction to enforce the foregoing direction, by contempt or otherwise.

Dated: May 29, 2003 Wilmington, Delaware

> BY ORDER OF THE BANKRUPTCY COURT Judith H. Wizmur, Bankruptcy Judge

# Exhibit C

Form of Notice of Entry of Confirmation Order

# IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

▝▋▋▙▙▖▖▖▖▖▖▖▖▖▖▖▖	X
In re:	•
LERNOUT & HAUSPIE	
SPEECH PRODUCTS N.V.,	:
	i
Debtor.	

Chapter 1]

Case No. 00-4398 (JHW)

# NOTICE OF ENTRY OF ORDER UNDER 11 U.S.C. § 1129 CONFIRMING THE OFFICIAL COMMITTEE OF UNSECURED CREDITORS' PLAN OF LIQUIDATION FOR LERNOUT & HAUSPIE SPEECH PRODUCTS N.V. <u>UNDER CHAPTER 11 OF THE BANKRUPTCY CODE</u>

TO ALL HOLDERS OF CLAIMS AGAINST AND HOLDERS OF EQUITY INTERESTS IN LERNOUT & HAUSPIE SPEECH PRODUCTS N.V., AND OTHER PARTIES IN INTEREST:

PLEASE TAKE NOTICE that, on May 29, 2003, the United States Bankruptcy Court for the District of Delaware (the "<u>Bankruptcy Court</u>") entered the Findings of Fact and Conclusions of Law Relating To, And Order Under 11 U.S.C. § 1129 Confirming, The Official Committee of Unsceured Creditors' Plan of Liquidation for Lemout & Hauspie Speech Products N.V. under chapter 11 of the Bankruptcy Code (such plan of liquidation, as modified and amended, being the "<u>Plan</u>," and such order being the "<u>Order</u>"). Capitalized terms not defined in this notice have the meanings ascribed to them in the Order or the Plan.

PLEASE TAKE FURTHER NOTICE that, subject to the occurrence of the Effective Date, the provisions of the Plan bind L&H NV, any person or entity acquiring property under the Plan, and any holder of a Claim or Equity Interest, whether or not the Claim or Equity Interest of such creditor or equity security holder is impaired under the Plan and whether or not such creditor or equity security holder has accepted the Plan.

PLEASE TAKE FURTHER NOTICE that, pursuant to the Plan, Priority Non-Tax Claims are unimpaired. Holders of Priority Non-Tax Claims will either be paid in full in Cash on the Effective Date or receive such treatment as to which L&H NV or Post Effective Date L&H, as the case may be, and such holder agree in writing. Unsecured Claims, PIERS/Old Convertible Subordinated Notes Claims, Common Stock, Securities Law Claims and Other Equily Interests are impaired and will receive only those distributions provided under the Plan.

Ö

PLEASE TAKE FURTHER NOTICE that, except as otherwise provided in the Plan or Confirmation Order (Including any right to receive Distributions under the Plan) or a separate order of the Bankruptcy Court, as of the Effective Date, all Entities that have held, currently hold or may hold a Claim or other debt or Liability that would be discharged or an Equity Interest or other right of an equity security holder that is terminated and canceled pursuant to the terms of the Plan, are permanently enjoined from taking any of the following actions on account of any such Claims, debts or liabilities or terminated and canceled Equity Interests or rights: (1) commencing or continuing in any manner any action or other proceeding against Post Effective Date L&H, the Chapter 11 Assets or properties and interests in properties of each of the foregoing; (2) enforcing, attaching, collecting or recovering in any manner any judgment, award, decree or order against Post Effective Date L&H, the Estate or properties and interests in properties of each of the foregoing; (3) creating, perfecting or enforcing any lien or encumbrance against Post Effective Date L&H, the Chapter 11 Assets or properties and interests in properties of each of the foregoing; (4) asserting a setoff, right of subrogation or recoupment of any kind against any obligation due to Post Effective Date L&H, the Chapter 11 Assets or properties and interests in properties of each of the faregoing; and (5) commencing or continuing any action, in any manner, in any place that does not comply with or is inconsistent with the provisions of the Plan and the Confirmation Order. In addition, all holders of Belgian Priority Claims shall be prohibited from asserting such claims in the Chapter 11 Case or seeking consideration pursuant to the Plan on account of a Belgian Priority Claim. Such injunction shall extend to all successors of the Debtor and its creditors. Nathing in the Plan or the Confirmation Order shall prohibit, impede or prevent any party from pursuing a claim or taking any other action in the Belgian Case, or from pursuing a claim or taking any action against any assets other than the Chapter 11 Assets.

PLEASE TAKE FURTHER NOTICE that, nothing in the Plan or the Confirmation Order shall be deemed to discharge, enjoin, restrict or otherwise impair any rights that may exist in favor of a person or entity to assert any defensive rights of setoff or recoupment with respect to any cause of action that may be asserted against such person or entity by the Debtor or successor in interest to the Debtor, including the Litigation Trust described therein.

PLEASE TAKE FURTHER NOTICE that, pursuant to the Plan and the Order, the satisfaction pursuant to Section 12.3 of the Plan shall also act as an injunction against any Entity commencing or continuing any action, employment of process, or other act against L&H NV or Post Effective Date L&H to collect, offset, or recover any Claim or Cause of Action satisfied, or released under the Plan. PLEASE TAKE FURTHER NOTICE that, pursuant to the Plan and the Order, in accordance with Section 7.4.2 of the Plan, and subject to Section 7.14 of the Plan, all of the Assigned Causes of Action that belong to or could have been raised by or on behalf of L&H NV or its respective Estate, shall be deemed transferred and assigned to the Litigation Trust. The assignment shall include, but is not limited to, that certain Cause of Action commenced by Kemper Indemnity Insurance Company against, inter alia, L&H NV bearing Civil Action No. 03-CV-2214 (S.D.N.Y.). Post Effective Date L&H and the Litigation Trust, as the case may be, as the successors to the Debtor, shall retain and may prosecute any of the foregoing as a defense or counterclaim to any Claim, Counterclaim or action, including, but not limited to, any rights under section 502(d) of the Bankruptoy Code. All other Causes of Action are being retained by Post Effective Date L&H in accordance with Section 7.14 of the Plan

PLEASE TAKE FURTHER NOTICE that, pursuant to the Plan and the Order, except to the extent of any willful misconduct or gross negligence, none of L&H NV, Post Effective Date L&H, or any of the L&H NV Parties (but solely in their capacities as L&H NV Parties) shall have or incur any liability to any holder of a Claim or Equity Interest for any act or omission in connection with, related to, or arising out of, (a) the Chapter 11 Case, (b) the pursuit of confirmation of the Plan, (c) the consummation of the Plan or the administration of the Plan or the property to be distributed under the Plan, (d) the Plan, (e) the negotiation, formulation and preparation of the Plan and any of the terms, settlements and compromises reflected in the Plan, and, in all respects, Post Effective Date L&H and each of the L&H NV Parties shall be entitled to rely upon the advice of counsel with respect to their duties and responsibilities under the Plan; provided, however, that the exculpation contained in this Section 13.2 of the Plan shall not apply, (x) with respect to the L&H NV Parties, to acts or omissions that occurred prior to the Plan Date, and (y) solely with respect to Post Effective Date L&H, the Litigation Trust, the Plan Administrator, the Litigation Trustee or the Curators, to acts or omissions that occur after the Effective Date.

PLEASE TAKE FURTHER NOTICE that, any party in interest wishing to obtain a copy of the Order may request such copy at their own exponse by contacting Colin M. Adams, Esq., Akin Gump Strauss Hauer & Feld LLP, 590 Madison Avenue, New York, NY 10022, at (212) 872-1000. Copies of the Order may also be reviewed during regular basiness hours at the United States Bankruptcy Court for the District of Delaware, 824 Market Street, 6th Floor, Wilmington, Delaware 19801, or may be viewed at the Bankruptcy Court's website, <u>www.deb.uscourts.gov</u>.

Dated: May 29, 2003 Wilmington, Delaware

> BY ORDER OF THE BANKRUPTCY COURT Judith H. Wizmur, Bankruptcy Judge

> > PATENT REEL: 019047 FRAME: 0223

**RECORDED: 03/20/2007**